学位論文



An Empirical Study of Protected Values and Public Acce ptance around Infrastructure Projects in Indonesia



愛媛大学大学院理工学研究科博士後期課程 生産環境工学専攻環境建設工学 コース

氏名 IRWAN SETIAWAN

2019年

CHAPTER 1

Introduction

1.1. Modernization and trends in infrastructure development in Indonesia

Indonesia has achieved significant modernization through the 70s to the 90s in the twentieth century which rendered the country into a member of second-generation NICs (Newly Industrializing Countries). The modernization in Indonesia was originally to emphasize the agricultural sector with efforts through intensification, diversification, extensification and verification of agriculture to lead food self-sufficiency (Asnawati, 2019). Then it was continued to improve the education, industry, trade, service, food technology, political system, defense, and security, as well as traffic facilities and infrastructure. Thus, Indonesian economy that was highly dependent on agriculture has changed into a more balanced economy in which the share of manufacturing (a type of industry) exceeds that of agriculture. This also implies that Indonesia lessened its traditional dependency on primary exports.

After economic reformation in modernization had been made, the other challenges and opportunities in Indonesia were to increase urbanization to provide greater prosperity and inclusion. To ensure that urbanization is able to work for all Indonesians, central and local governments need to cooperate in establishing policies that achieve some objectives. Some of the purposes are to integrate between places, cover investing in improved transportation networks and policies and leverage the supply of affordable urban housing. Integration promotes mobility and generates inclusiveness.

Although the modernization has no direct impact to developing infrastructure, it has greatly improved the "hard and soft" infrastructures. The hard infrastructures are those such as roads, airports, bridges, houses, electrical power, and garbage disposal facilities, and the soft ones are those such as social welfare, healthcare, and participation in infrastructure programs or projects. Progress has been made by several objective standards. A number of high-profile infrastructure projects are being completed, particularly in Java and Sumatera islands. New reformations which have been made such

as infrastructure investment need to be much improved by the business climate and by deeper tax reform.

In year 2017, the Public Works and Public Housing Ministry plans to focus on building infrastructure in four sectors next year, namely transportation, food supply, waste management and housing. The ministry is set to obtain Rp 106.9 trillion from the proposed 2018 budget once it is approved, the highest amount of all ministries and government institutions. Of that figure, Rp 41.3 trillion would be allocated to building new roads and bridges as well as maintaining existing ones, while another Rp 37.3 trillion was earmarked for the construction of dams and irrigation networks, said Adang Saf Ahmad, a member staff of the Public Works and Public Housing Minister.

1.2. Democratization and citizen participation in Indonesia

Indonesia's democratization was initiated in the early 1980s. The *orde baru* government was ready to turn its attention to a series of economic reformations that slowly transformed the economy and laid the basis for a decade of robust growth (Gyda, 2012). Besides economic reformion, the bureaucratic capitalism system, which existed in the country, was strengthened during the years of Suharto; there were many reforms in policies, both military and business. Participation of citizens in Suharto's period was very low, most people only followed the government's will and almost all development projects were implemented by the government using top-to-bottom systems. Furthemore it took almost ten years to achieve the expected level of development. It was contrasted to the scholarly critique of the deficits of Indonesian democracy since the transition to democracy in 1998, focusing on links between democracy and civic agreement as indicators of a flourishing democracy, evident in the general- and specific-context literature on democracy (Gyda, 2012). Democratic reforms and the elites had changed and been allowed to maintain the integrity of the country and give encouragement to the transition of bureaucratic capitalism to oligarchic capitalism. The literature on participatory development makes similar kinds of assumptions regarding the effect of popular and civic engagement on democracy. For example, the societies' participation in the

process of development planning and decision-making provides an arena for bottom-up democratic governance which was noted from a World Bank report (Guggenheim et al., 2004). At the local level, civic engagement will enable the creation of trust-networks and cooperation that strengthens a community's social capital, thus creating necessary relationships among people who work in government and people in civil society (Putnam et al., 1993).

The fundamental value of community participation in Indonesia should not be understood as an entirely new trend. During the period, local communities and villages in Indonesian social structure were identified, streamlined and reformed to represent 'culture' and 'tradition' (Sidel, 2007, pp.31-32). The ideas of *gotong royong* (mutual aid) and *musyarawah* (rule by consensus) not only figured prominently in ethnographic accounts of the Indonesian village (desa) and urban neighborhoods (kampung), but were used to abbreviate the New Order bureaucracy from the state down to the village and neighborhood level (Antlöv & Cederroth, 1994; Pemberton, 1994, pp.238-39).

In the last decade, the launching of infrastructure packages, and the introduction of regulatory and institutional reformation meant to attract public private partnership (PPP) in infrastructure and the participation level of citizens on project development has changed and become more common, such that the citizens can get involved by sharing their ideas for projects in their own area; hence the system is turning into a bottom-up one (Gyda, 2012). In the government regulation number 25/2004 on National Development Planning, the structure of community participation through *Musrenbang* was first formally endorsed (Gyda, 2012). Musrenbang is the term used for the development planning and multi-stakeholder consultation forums which are meant to encourage and promote community participation in development planning at the regional level. A citizen's representative is invited to a formal meeting with local government, a project officer and a public-private partner to discuss the projects around the area.

The opportunity of citizen participation in decision-making in infrastructure is increasing over that in the Suharto period. Previously, people were passive in decisions about infrastructure projects. They could not give any input or ideas concerning their needs; it was almost impossible to suggest anything to the government. However, recently that condition has changed. The government formed a new program and reformation democracy, in which people can get involved directly in the process of decision-making for the projects. Participation in decision-making, planning, and budgeting was thus as much a demand of the citizens as a result of shifts on official development discourse away from technocratic top-down implementation of development programs towards bottom-up participatory models of development (Sarosa et al., 2005).

The opportunities of citizen participation in decision-making for infrastructure have been mentioned above. It starts with the idea that *gotong royong* (mutual aid) and *musyawarah* (rule by consensus) not only figure prominently in ethnographic accounts of the Indonesian village (desa) but also in urban neighbourhoods (kampung). This is a kind of culture and tradition in local villages, where all people are involved not only in coming up with ideas, but also in working together. For example, if a village needs to build a bridge to connect to a neighborring village, then the citizens will hold a meeting where many may share their thoughts, as well as support the implementation of this project by bringing supplies for the people who will work on the project. They will bring food, water, and other things to help until the bridge is finished.

1.3. Public acceptance of infrastructure projects

Public acceptance is an important concern for multiple reasons (Bicket et al., 2016). For instance, perceived public acceptability, especially the perception of controversy, is itself also a key factor in the public acceptability of a policy; people are less likely to support a policy if they expect others to defect (De Groot et al., 2012). A strong social norm, where attitudes and behavior in support of the policy are common knowledge, is associated with higher public acceptability (Schade et al., 2003). Perceived distributional fairness of costs and environmental burdens (Dresner et al., 2006; Zverinova et al., 2014), and procedural fairness (Drews et al., 2015) are linked to higher public acceptability. Increasing public acceptance is considered to be an important factor in designing the

5

decision-making process, especially regarding infrastructure construction. If the public perceives the project to be fair in procedure and in cost-benefit analysis, then they will accept it.

This research focused mainly on two infrastructure projects to investigate public acceptance: one was construction of an elevated road and the other was a facility for disposal of garbage. In the first case, the government promised that the construction of the elevated road would prevent traffic jams, which were commonly experienced. However, cemeteries, mosques, residences, and public facilities such as traditional markets and schoolyards had been moved or destroyed for this project. The residents are angry about this project. The project is currently postponed until the residents give their permission for it to continue.

Thus, public acceptance is a key factor in the successful implementation of regional policies (Wan, 2017). If the public broadly accepts a project proposal, government will be able to implement it smoothly. Gaining support from residents may also help authorities raise funds for the project (Basbas et al., 2015). On the other hand, low levels of acceptance among residents may create obstacles to the progress of a project despite the best efforts to implement it (Payne et al., 1996) As a result, a project may not be implemented because of strong public opposition.

1.4. Social conflict around infrastructure projects

Many cases of social conflict occur in infrastructure projects. The land acquisition is an extremely problematic phase in infrastructure development. As mentioned above, the case of building garbage disposal facilities is one of the cases that indeed has social conflict around land acquisition. The project was needed because residents produce garbage daily, dispose of their garbage anywhere, and then the government has to collect it all and bring it to the disposal facility. The government needs more space for this facility. But, the citizens around the facility have strong opinions and tend to disagree with this project. In this circumstance, the government needs support from the residents and other stakeholders. They should approach and invite the residents to a meeting to discuss alternative solutions. The government needs to build consensus to solve this conflict. However, there

are many cases in which a project is rejected because of strong public opposition. Especially, people with strong opinion prevent consensus building.

Consensus building is an especially interactive shape of a participatory process in which the affected stakeholders are directly involved in the decision-making (Dorcey et al., 1994). There has been an increasing interest in involving the public in decision-making and community development, both in the developed and developing world (Rowe and Frewer, 2000) in the last decade. At the same time new participative planning techniques focusing on consensus building have come up (Godschalk and Patterson, 1999) in the last decade. Many researchers have studied consensus building related to procedural and public acceptance, but few research studies have been made about the relation between procedure and absolute values that are protected on decision processes accepted by the public.

1.5. Protected values

Regional projects are commonly constrained by various trade-offs between incompatible values, as satisfying one value may entail sacrificing another. For example, the studied project of the highway can improve regional accessibility and reduce travel time, but it can also lead to changes in the local environment. Theories of rational decision-making require trade-offs among values (Baron and Sprance, 1997).

It has been pointed out, however, that residents may have strong feeling about mosques, residences, and public facilities such as cemeteries, traditional markets and schoolyards being moved or destroyed for a project. When people have values and are against trade-offs with other values, this is generally called protected values by Baron and Spranca (1997), or an attitude where one will not compare or balance particular important values with other values. In other words, these values are based on an absolute belief that one's values must be protected at all costs. People may assign protected values to human and animal life, the natural environment, human rights, divinity, etc. People who have these values are difficult to change or trade-off to other values, especially for a project. Moreover, some projects displace neighborhood residents. As long as residents' responses to

7

proposed projects either directly or indirectly affect regional decision-making regarding the project, residents need to understand such trade-offs to make well-reasoned judgments about the projects.

1.6. Purpose of this research

This study uses social psychological approach to understand the issues of consensus building or residents' acceptance for public works, especially focusing on the association of protected value and procedural fairness in the field of social psychology.

- 1. To explore the impacts of protected values on consensus building (or acceptance) around infrastructure projects.
- To explore the association between protected values and procedural fairness in decision processes for infrastructure projects.
- 3. To develop an effective policy measure to mitigate the negative impacts of protected values in order to promote appropriate decision-making for infrastructure projects.

1.7. Structure of the Dissertation

This research is divided into seven chapters. Following are the description of each chapter:

Chapter-1

This section explains: modernization and trend in infrastructure development in Indonesia, democratization and citizen participation in Indonesia, public acceptance of infrastructure projects, social conflict around infrastructure projects, protected values, purpose of this research and structure of this dissertation.

Chapter-2

This section defines the reference that cited accordingly to the discussed subject in this research, determinant of acceptance, properties of protected values, procedural fairness, the focus of present research.

Chapter-3

The surveys explained in this section focuses on the household to be displaced and discussed about the influential factors on public acceptance. This chapter is divided into three parts. Part one presents trade-off judgment of relocation and protected values and the present hypotheses. Part two describes the study area and respondents. Part three describes questionnaire, results and discussion.

Chapter-4

The survey explained in this section focuses on the impact of protected values on acceptance and procedural fairness on public projects. This section is divided into four parts. Part one presents the introduction of this research. Protected values, protected values and decision procedures. Part two describes about method of the research, the participants, questionnaire and procedure. Part three the results part four is discussion.

Chapter-5

This section describes about the protected values and their influences on perceived procedural fairness in highway project. Therefore, the highway construction project is to explore the nature and prevalence of protected values relevant to the project. Then, the study explored the role of trust in reducing the detrimental effects of protected values. This section is divided into six parts. Part one presents the introduction of research. Part two describes theory about property of protected values and procedural fairness. Part three describes methods and description of flyover construction project and respondents. Part four describes procedures and instruments. Part five is results showing possession and properties of protected values, impacts of protected values on procedural fairness, effect of trust in government and structural equation model analysis and part six is the discussion.

Chapter-6

This section describes about the effect of providing opportunities for reflection regarding Protected values (PVs) on the changes of the PVs and self-assessment of understanding. The issues of forming consensus about taking significant community risk, such as the construction of high-level radioactive waste in city, incorporates a trade-off between compensation (benefits) when the risk is accepted and values impaired. This section divides into four parts. Part one is present the introduction, the concept of protected values and protected values and the illusion of understanding and hypothesis of research. Part two is describe experiment participant and procedures. Part three is the result about tendency to hold protected values and associated psychology characteristics, change in protected values and assessment of understanding, and relationship between protected values and assessment of understanding. Part four is the discussion about implication for the issue of consensus formation and subjects for future investigation.

Chapter-7

This section describes about the conclusion of the study and its implication to public works in Indonesia.

CHAPTER 2 Theoretical Background

2.1. Determinants of acceptance

Psychological acceptance is defined as one of the most important contextual changing strategies (Hayes, 1994). It means that acceptance refers to the conscious abandonment of a direct change of agenda in the key domains of private events, self, history, and openness to experience thoughts and emotions as they are. In this same vein, Dougher (1994) suggests that the key component of acceptance is letting go of one's control agenda and orienting towards valued actions. In other words, that acceptance is not an end goal but it's a method of empowering gain of the life goals. If public acceptance is the key to the successful implementation of regional policies (Wan C., 2017), then the public is able to broadly accept a method of project proposal which empowers the goal, governmental authority. As a result, a project may not be implemented in the public opposition.

The psychological research on social acceptance says that it has been generally important to note that this is a distinct type of acceptance, not to be confused with the common use of the term acceptance with respect to the status quo of one's life situation (Hayes, 1994). On the other hand, when policies are perceived as unacceptable the mechanism will work the other way, resulting in higher perceived costs and lower perceived benefits associated with the policy (Siegrist, Cvetkovich, & Roth, 2000). Hence, perceived costs appeared as the one of key predictors that strongly reduce the acceptance of policies (Bronfman, Jiménez, Arévalo, & Cifuentes, 2012). Frewer, Howard, & Shepherd (1998) assumed that acceptance of the policies is affected by perceiving costs and benefits that are functionally related to each other. Changing perception of benefits may be possible to alter perceived cost. This means that if people tend to be consistent in their beliefs, this results in devaluation of costs and the elevation of benefits for policies perceived as acceptable.

Residents' response to the government's proposed project depends on their perceptions of its benefits and costs, and the trade-offs between them (Hamersma, Heinen, Tillema, & Arts, 2016). Regional projects are commonly constrained by various trade-offs among incompatible values, as satisfying one value may entail sacrificing another. For example, transportation projects such as highway construction are able to improve regional accessibility and reduce travel time, but they can lead to change in the local environment and force local residents to relocate as well. If residents' response to proposed projects is taken into account directly or indirectly in regional decision-making, it is vital that residents figure out the trade-offs thus they can make well-reasoned judgments about the project. They also stated that fairness is one of the important values, which secures the present and future social welfare. Moreover, Fujii (2017) stated that if any policies greatly enhance social benefits, there is the highest possibility that people will regard the policies as fair. This implies the possibility that when the expectation is to increase the public benefit it fosters fairness through which public acceptance is increased. The acceptance of the public on regional projects will be studied in this research to measure the level of public participation and acceptance before and after understanding the projects.

2.2. Properties of protected value

Protected values are generally defined as a set of values protected from trade-offs with other values (Baron and Sprance, 1997). Most people have some values that they think as absolute ones, not to be traded-off for anything else. For instance, it has been pointed out that people may assign protected values to human and animal life, natural environment, human rights, divinity, etc. So those people will not condone any activities of development that could bring about extinction of plants or animals or acts of buying or selling organs for profit (Baron, 2008; Baron and Spranca, 1997). In other words, these values are based on an absolute belief that one's values must be protected at all costs or an attitude in which one value will not be compared with other particularly important values.

Baron and Spranca (1997) reported that a protected value is based on the *deontological rule*, which contrasts with consequentialism and represents the normative position that judges the morality of an action based on its adherence to a rule or rules. Unlike consequentialism, protected values are considered to be based on deontological rules regarding behavior itself rather than the consequence of the behavior (Baron and Spranca, 1997). Here, deontological rules generally represent that governed actions should or should not be taken, regardless of their consequences. According to Baron and Spranca (1997), protected values may show qualities of: 1) "quantity insensitivity," 2) "obligation," 3) "anger," and 4) "posturing" as based on such deontological rules (Baron and Spranca, 1997).

Properties of protected values can be described as duty-, obligation-, or rule-based beliefs because these beliefs stipulate conformation of people to certain behavior standards. Actions are more important than consequences for people who possess protected values. They have a belief in obligating support or non-support for a certain action, regardless of the result of the action. Baron and Spranca (1997) identified four characteristics of protected values derived from the standpoint of the deontological rule: moral obligation, quantitative insensitivity, anger, and absoluteness. Firstly, protected values are observed as moral values that are universal and objective, regardless as a moral obligation it is not simple conventions or personal preferences. Secondly, people's insensitivity may be caused by protected values. For instance, an environmentalist with protected values considers destroying a species through a single act to be as bad as destroying a hundred species through a single act. Thirdly, if people's protected values are violated then they may become angry because they observe it as a moral violation. Fourthly, absoluteness can be associated with protected values, as people believe that these values should be protected from any trade-offs under any circumstances. This study examined these properties that could be observed among Indonesians by their attitudes toward a regional project.

2.3. Procedural fairness

The perceptions of fairness, that consist of procedural and outcome fairness, are crucial in facilitating residents' acceptance of policies (Lind & Tyler, 1988; Thibaut & Walker, 1975; Tyler & Lind, 1992). Psychological theories of justice assume that people care about justice and fairness primarily because of their social identity needs (Skitka & Mullen, 2002). People may be particularly attentive to procedural fairness because of procedural actions by institutions and authorities which provide important information about the social worth and values of the involved parties (Cropanzano & Greenberg, 1997). Procedural and outcome fairness are conceptually independent but normally they have correlation. They can even be important tools in observing policy which can be perceived as fair depending on the situation and institutional framework of a given country (Drever & Walker, 2013). Procedures are posited to be the dominant predictor of the perception of outcome fairness and acceptance because people are usually aware of procedures before outcomes. According to Lind and Tyler (1988), belief about procedural fairness is the strongest predictor of whether people feel justice was done in a society. Tyler et al. (1996) reported that giving disgruntled group members a voice, regardless of whether it is instrumental (i.e., voice can affect the decision-making process) or noninstrumental (i.e., voice has no effect on the decision-making process), may be sufficient for a process to be considered fair. In addition, Wenzel (2002) shows that if people feel fairly treated, they will be satisfied with less favorable outcomes.

Perceived benefits have significant influence on procedural fairness and outcome fairness. Residents tend to have higher perceptions of perceived fairness, both procedural and outcome fairness. This implies that increasing the perceived benefits of the policy is sufficient to make residents recognize the policy as fair, which may foster public acceptance. This new fact is in line with Garling, Fujii, & Jakobsson (2003) stating that increasing perceived benefit which consists of individual benefit and public benefit does not enhance the perceived fairness. Contrary to perceived costs having a small influence on acceptance, procedural fairness and even on outcome fairness have no influence. However, perceived costs have significantly influenceed the perceived benefits in a negative way and those have direct effect on perceived fairness. In the other words, it was said that perceived costs have indirect effect on both procedural and outcome fairness.

Frewer, Howard, & Shepherd (1998) assumed that an acceptance of the policies is affected by perceived costs and benefits that are functionally related to each other. They also argued that it may be possible to change perceived costs by changing perceptions of benefits. If people prefer consistency among their beliefs, this results in devaluation of costs and the elevation of benefits for policies perceived as acceptable. For policies perceived as unacceptable the mechanism would work the other way, resulting in higher perceived costs and lower perceived benefits associated with the policy (Siegrist, Cvetkovich, & Roth, 2000). Hence, perceived costs appeared as the one of key predictor that strongly reduced the acceptance of policies (Bronfman, Jiménez, Arévalo, & Cifuentes, 2012).

If study is continued on perceived benefit which is related to perceived fairness, then it is possible to expect an increasing of acceptance. Moreover, according to Schwartz & Bardi (2001) perceived benefits are also related to the perceived fairness. Residents have taken the perceived benefits, both individual and social benefits into account considering the policies' fairness. They also stated that fairness is one of the important values which secure the present and future of social welfare (social benefits). Moreover, Fujii (2017) stated that if any policies greatly enhance social benefits, there is a high probability that people will regard the policies as fair. This implies the possibility that expectation to increase the public benefit fosters fairness in which public acceptance is increased.

Trust in the government (such as regulators or owners of the policy) was found to be positively related to the residents' acceptance of policies (Siegrist et al., 2000). When residents have lack of information about a certain policy, acceptance may depend on trusting to the government who is responsible for the policy as a heuristic or alternative ground to base one's opinion (Siegrist & Cvetkovich, 2000). However, there is no agreement about the certain definition of trust and types of

15

trust. Residents who trust in the government more likely have to trust that the policies are effective as well as less harmful and fair, thus their acceptance tends to be higher (Schmöcker, Pettersson, & Fujii, 2011b). Trust in the government is based on whether among the residents the government is seen as having similar values or understanding to the specific situation (Earle & Cvetkovich, 1997). Government is expected to have similar views and values as the residents as well as making sure that the residents obtain the proper benefit from any policies they made. As a result, it is expected that the fairness is perceived as significantly high when the benefits perceived by residents is higher.

Thus, a regional project tends to be accepted when the process surrounding the project is considered fair. After recognizing that a project conflicts with their protected values, residents might not feel that the procedure is fair. Particularly according to the value protection model developed by Skitka (2002), people are motivated to protect their sense of personal identity when it is threatened, and they perform this by making cognitive, affective, and behavioral adjustments. All of which suggest whether they will feel an event is fair or unfair.

2.4. The focus of present research

Although previous research has studied protected values in an extensive manner, several limitations require consideration. First, few studies have dealt with this issue in the context of regional public policy-making. Furthermore, previous studies by Baron and his colleagues have explored protected values using hypothetical scenarios. Little is known about whether people have protected values when it comes to actual projects, and, if they do, how such protected values affect acceptance of the projects. Second, previous studies have emphasized the importance of procedural fairness in residents' acceptance of public decisions (Lind & Tyler, 1988). Yet, it is finding that residents' perceptions of procedural fairness can be affected by the presence of protected values. Many research projects were limited to measuring the protected values regarding public acceptance, and to determining if decision procedures can be used to moderate negative attitude, and to examine whether

people with protected values perceive the fairness of the decision procedures. There is not yet a study about the association between procedural fairness and protected value. Finally, few studies have explored ways to mitigate the negative impacts of protected values on public acceptance around infrastructure projects.

Given these limitations, this research addresses the problematic characteristics of protected values that impede public acceptance, focusing on actual projects. Then, it explores the association between protected values and procedural fairness of decision processes for infrastructure projects. Finally, an effective policy measure to mitigate the negative impacts of protected values in order to promote appropriate decision making for infrastructure projects is examined and proposed.

CHAPTER 3

The Effects of Protected Values on Policy Acceptance

3.1. Introduction

One of the main challenges to the implementation of residential relocation as a part of urban renewal policy is the need to gain acceptance from residents. Even though relocation of residents is necessary as a means of urban restructuring such as industrial development, slum clearance or disaster prevention, it is not always welcome by the affected residents (McDonald-Wilmsen & Webber, 2010; Mehta, 2005). The government proposal of relocation, both voluntary and involuntary or "forced" programs, can fail to be carried out in the face of strong opposition from residents. Therefore, to understand determinant factors underlying residents' acceptance is a key to effectively implement relocation programs.

3.1.1. Trade-off judgment of relocation and protected values

Numerous studies have viewed household's decision to relocate as a complex function of trade-offs process between current residence and relocation site (Heaton, et al., 1979; Kim, et al., 2005). A relocation program proposed by governmental authority commonly comes with various trade-offs with regards to a wide range of attributes such as housing costs, dwelling quality, accessibility, amenity conditions. On the one hand, a relocation program, especially as seen in the case of resettlement programs for slum clearance in Asian cities, could bring benefits to residents, for example when they come with rehousing to new dwelling (Yuen et al., 2006). On the other hand, residents may have a concern about the burden of additional costs due to change in their residence such as increase in housing costs, worsening of work conditions and social segregation (Stephens, 2010). Given this, it may be considered that residents are likely to accept the relocation program if they feel the benefits outweigh the costs.

Many studies on decision theory, however, have claimed that people are not solely concerned about the consequences of their own decision, nor make a trade-off between its benefits and costs (Baron, 2008). Assuming that's the case, even if a relocation proposal stresses the benefits that result from the relocation, the affected residents may reject the proposal. Indeed, recent studies suggest that residents' refusal to leave was still observed despite governmental efforts to mitigate their hardship through compensation and quality of life improvements in the new area (Dirar et al., 2015). Thus, residents' responses to the government proposal of relocation can be influenced by other factors which are at least somewhat independent of its consequences.

The present study focuses on a decision rule to resist trade-offs, which is known as *protected values*, as an important factor preventing affected residents from accepting a relocation program whatever its consequences. Baron and Spranca (1997) called values that are protected against trade-offs with other values protected values. Using economic terminology, protected values are values with an infinite marginal rate of substitution. A protected value is based on deontological rules, which contrasts with consequentialism and represents the normative position that judges the morality of an action based on its adherence to a rule or rules (Baron & Spranca, 1997). People with protected values think that these values should not be sacrificed for anything, regardless of the benefits. When residents possess such a strong value against being relocated, they can hardly make a trade-off between perceived benefits and costs of the relocation. Accordingly, they will keep rejecting a relocation proposal, whatever benefits the government offers to them. Therefore, residents' denial of trade-offs due to loyalty to their protected values may challenge authorities' attempts to garner public acceptance of residential relocation.

The present study is aimed to demonstrate the effects of protected values on residents' acceptance of relocation programs by examining a case of relocation programs of riverbank settlers in Jakarta, Indonesia.

3.1.2. The present hypotheses

In addition to protected values, residents' acceptance of relocation program may also depend on two other factors not directly related to consequences of the relocation (Kearns & Mason, 2013). First, the strength of place attachment within their current residence has been shown to promote residents' willingness to stay the location and therefore may be negatively related with their acceptance of the relocation (Fischer & Malmberg, 2001). Second, trust in government has been recognized as a crucial factor that influences individual's acceptance of the government policy. Both factors may also influence aforementioned factors, i.e. perceived costs and benefits and protected values. It is because the costs and benefits of relocation may become higher and lower for residents who have more place attachment, and it can be considered that such residents then refuse to make a trade-off between the costs and benefits. Also, if residents do not trust in the government, they are likely to perceive negative outcomes from its relocation proposal. In addition, it is suggested that government trust is a root cause of people's protected values against the government policy.

The conceptual framework is shown in Figure 3.1. The assumed causal relations are indicated by the presented arrows. Coefficients to be estimated express the strength and sign of these paths. In the estimation, hypothesized latent variables correspond to the theoretical constructs which in turn are related to the observed variables through measurement models.



Figure 3. 1.A theoretical model of determinants of acceptance of relocation program

3.2. Method

3.2.1. The study area and respondents

The research site chosen for this study was Bukit Duri, a village (kelurahan) located on the Ciliwung River in South Jakarta (Figure 3.2. and 3.3.). The district, covering approximately 1.08 km2, is a densely populated area in which 41,938 people live in one- or two-story dwellings (Figure 3.4.(a) and 3.4.(b)). They have experienced floods (Figure 3.4.(c)) almost every year; the highest inundations on record were 7 m in 2007 (Antara News, 2012) and 4 m in 2013 (Vollmer, Prescott, Padawangi, Girot, & Grêt-Regamey, 2015). The government claims that clearing the informal settlements by displacing the residents to new settlements will improve the city's environment and solve its flooding problem (van Voorst, 2016). She then plans to displace around 728 households in Bukit Duri into housing that is located approximately 15 km away from the current settlement as flood mitigation measures called the River Normalization Program. The affected residents, however, refused the government's proposal, saying the eviction process was unfair and violated their human rights. The government's forced eviction invoked several rallies from residents.



Figure 3. 2 Map of Jakarta province



Figure 3. 3 Research area



Figure 3. 4.(a) Housing conditions, (b) riverbank settlements, and (c) post-flood inundation damage

Data collection for this study was conducted during September and October 2016. Two hundred and fifty respondents were selected by simple random sampling from 728 households that were directly affected by the displacement. This research used a quantitative method that included data collected from household surveys and structured questionnaires. Of the 250 returned questionnaires, one questionnaire was excluded due to incomplete answers, leaving 249 usable responses.

3.3.Questionnaire

Table 3.1. presents an overview of questionnaire items used in this study to measure five variables: acceptance, perceived cost and benefits, protected values, place attachment, and trust. Following Kim, Schmocker, Bergstad, Fujii, & Garling (2014), respondents' acceptance of relocation was measured using three items. Perceived benefits and costs were measured with two items respectively, which assessing the residents' perception of the positive and negative outcomes that might result from the relocation policy. As for protected values, two items were used to measure the

degree to which residents resist trade-offs with being relocated. Place attachment was measured with four items, which were taken from Hernández et al. (2007). Finally, trust in the government was measured with two items, one assessing general trust and the other assessing trust with regards to the relocation program. All of the questionnaire items were measured using a 7-point Likert scale, with the higher scores indicating greater agreement with the questions or statements.

Table 3. 1. Questions used to measure each variable					
Acceptance of	Do you agree with government policy regarding relocation? (A1)				
relocation program	Do you support the government policy regarding relocation? (A2)				
	Are you willing to accept this government's decision to implement the				
	relocation policy? (A3)				
Perceived benefits	How improved the condition of the new settlement's buildings do you				
	expect compared to your current residence? (PB1)				
	How improved the public school condition in the new settlement do you				
	expect compared to your current residence? (PB2)				
Perceived costs	Do you think the moving cost for the new settlement is high? (PC1)				
	Do you think you will find difficulties and spend more money for				
	changing your lifestyle in the new settlement? (PC2)				
Protected values	It is impossible for me to think about how much benefit we should				
	demand in order to allow this displacement. (PV1)				
	There is nothing we can gain by following this displacement. (PV2)				
Place attachment	When I've been away for a while, I really want to come back. (PA1)				
	I feel at home in this neighborhood. (PA2)				
	When I'm away, I always miss my house and my neighborhood. (PA3)				
	This neighborhood is part of my identity. (PA4)				
Trust in	In general, do you trust the government? (T1)				
government	Do you trust the government to make policy on relocation? (T2)				

3.4.Results

To examine the comprehensive causal relationships among five variables, an SEM based on path analysis including the five variables as latent variables is used in AMOS. Figure 3.5. shows how the latent variables were estimated from the measurement variables described in the Section 2. The measures labeled A1-A3, PB1-PB2, PC1-PC2, PV1-PV2, PA1-PA4 and T1-T2 refer to the ratings of acceptance, perceived benefits and costs, protected values, place attachment, and trust, respectively. As shown in Table 3.2., all estimates of parameters in the measurement models were significant. This suggests that the observed variables are measures of the theoretical construct as hypothesized.



Figure 3. 5. Estimated model of determinants of acceptance of relocation program. (The estimated coefficients are given for each causal path in the structural model accompanied by the corresponding t-statistics within parentheses.)

Latent variables	Measurement variables	urement variables Coefficient	
Acceptance of	A1	A1 1.00	
relocation program	A2	A2 1.13	
	A3	1.02	16.58
Perceived benefits	PB1	1.00	-
	PB2	1.01	9.13
Perceived costs	PC1	1.00	-
	PC2	0.93	7.78
Protected values	PV1	1.00	-
	PV2	1.46	4.91
Place attachment	PA1	1.00	-
	PA2	0.89	11.54
	PA3	1.06	14.12
	PA4	1.01	13.91
Trust in	T1	1.00	-
government	Τ2	0.94	24.91

Table 3. 2. Estimates of coefficients in measurement models

The estimated coefficients corresponding to the hypothesized causal paths between the latent variables are also given in Figure 3.5. In support of the hypotheses, the coefficients corresponding to the posited causal paths, except for the paths between place attachment and perceived benefits and between place attachment and protected values, were significant with the right signs. In addition, the model fit the data well, as judged by the following statistics: χ^2 (n = 249, df = 80) = 139.65, $\chi^2/df = 1.75$, CFI = 0.97, AGFI = 0.90, and RMSEA = 0.055.

3.5. Discussion

The results of the present study indicate that protected values tend more to affect negatively residents' acceptance of relocation program than their perception of benefits and costs of the relocation do. This implies that residents may have concerns not solely about the consequences of the

relocation. The findings of this study suggest that the government may not able to gain residents' acceptance simply by offering compensation and new settlements, especially in case that the residents with protected values refuse to calculate whether the benefits of the relocation exceed the costs

It was also shown that protected values were negatively related with trust in government. The association between protected values and trust has not yet been examined in previous research, but is in line with the arguments in Baron & Spranca (1997) suggesting that those who distrust others think that allowing trade-offs would be too risky. Given the present findings, the government should make efforts to promote trusting relationships with residents through, for example, showing their sincerity. In addition, acceptance was shown to be negatively related with place attachment. This result was consistent with the findings of Fischer and Malmberg (2001). It would be important that the sense of place or community be taken into consideration and maintained even in case of relocation.

Further research should include various factors, such as risk perception and procedural fairness, to more comprehensively understand residents' attitudes toward relocation. Psychological processes, including those factors that underlie residents' acceptance of displacement, could be examined in a more systematic way by applying structural equation modeling. Furthermore, it is also important to study the psychological processes by which protected values are developed or mitigated. Experimental research that examines the effects of reflective thinking and communication processes on protected values would contribute to developing relevant measures to reduce the negative impacts of protected values.

CHAPTER 4

The Impact of Protected Values on Acceptance and Procedural Fairness of Public Projects: Exploring the Effects of Decision Procedures

4.1. Introduction

Gaining acceptance from the public plays a critical role in the successful implementation of regional projects (Wan, Shen, & Choi, 2017). If a project proposal enjoys strong public support, the government can carry it out smoothly. Support from citizens may also support the authority in raising funds for the project (Basbas, Mintsis, Taxiltaris, Roukouni, & Vazakidis, 2015). Conversely, low levels of acceptance among citizens may cause obstacles to policy-goal achievement, despite a government's best efforts. As a result, a project may fail in the face of strong public opposition.

Residents' responses to government project proposals depend on their perceptions of its benefits and costs, and the trade-offs between them (Hamersma, Heinen, Tillema, & Arts, 2016). Regional projects are commonly constrained by various trade-offs between incompatible values, as satisfying one value may entail sacrificing another. For example, transportation projects such as highway construction can improve regional accessibility and reduce travel time, but they can also lead to changes in the local environment and force local residents to relocate. If residents' responses to proposed projects are taken into account, directly or indirectly, in regional decision making, it is vital that residents understand such trade-offs so that they can make well-reasoned judgments about the project.

4.1.1. Protected values

Some people with strong values and opinions, however, think that their values and opinions are absolutely non-negotiable and refuse to make trade-offs. Such values, which are protected against trade-offs with other values, are called *protected values* by Baron & Spranca (1997). Using economic terminology, protected values are values with an infinite marginal rate of substitution. People with

protected values think that these values should not be sacrificed for anything, regardless of the benefits. It has been pointed out that people may assign protected values to human and animal life, the natural environment, human rights, divinity, etc. so as not to condone any activities or developments that could lead to the impairment of such values (Baron, 2008; Baron & Spranca, 1997).

Unlike consequentialism, protected values are considered to be based on deontological rules regarding behavior itself, rather than the consequence of behavior (Baron & Spranca, 1997). Here, deontological rules generally represent rules that govern actions that should or should not be taken, regardless of their consequences. Protected values exhibit qualities such as quantity insensitivity, obligation, anger, and omission bias because they are based on such deontological rules (Baron & Spranca, 1997). First, protected values make people insensitive to consequences. For example, an environmentalist with protected values linked to environmental conservation may consider destroying a species through a single act to be as bad as destroying a hundred species through a single act. Second, the actions required or prohibited by protected values are regarded as moral obligations, because they are universal and objective, not simply conventions or personal preferences. Third, people may become angry if their protected values are violated because they see it as a moral violation. With regard to this, Tetlock et al. (2000) pointed out the possibility that the very act of trading off one's values with other people's values may induce a sense of anger. Finally, protected values concern actions but not omissions. Accordingly, people who hold protected values tend to be less concerned about the harm caused by omission than about identical forms of harm caused by action.

In the decision-making process for regional projects, such refusals to make trade-offs by some stakeholders create problems for government agencies, which try to allocate resources by taking into account the various values of all stakeholders. For example, one stakeholder could dominate a decision by expressing an absolute value, or stakeholders with conflicting protected values could make it impossible to finalize a decision (Baron & Leshner, 2000). Taken together, residents' denial

of trade-offs due to their protected values may challenge authorities' attempts to make reasoned judgments and garner public acceptance of decisions in diverse societies.

4.1.2. Protected values and decision procedures

Protected values are based on an absolute belief that one's values must be protected at all costs. Previous studies, however, suggest that such values may not necessarily be invariable; they may be altered, depending on the situation. Baron & Leshner (2000) demonstrated that protected values may be attenuated in those holding them when they imagine counterexamples to those values (circumstances in which they had no choice but to accept the action they object to). Similarly, Hatori & Kajiwara (2014) found that protected values may not be retained when people are given the opportunity to reflect on situations in which their protected values conflict with other values. These findings suggest that, even if people originally think that a project proposal conflicts with their protected values, they can, eventually, accept it.

Taking these findings into consideration, the present study focuses on the effects of decision procedures to explore the possibility of convincing those who have protected values to accept a public project. Given that protected values are based on deontological rules, which contrast with consequentialism, even if a project may lead to "consequences" that conflict with one's protected values, negative attitudes towards that project might be moderated, depending on the "procedures" used to design the project. According to Fujii (2008), methods for making social decisions can generally be divided into four types; 1) cost-benefit analysis, 2) compliance with laws and customs, 3) majority rule, and 4) deliberation. People's responses to a project proposal may depend on which method the government adopts to make the decision. In particular, as cost-benefit analysis measures trade-offs between the costs and benefits of a project in monetary terms, people who think their values should not be traded off may refuse projects proposed based on such a procedure.

4.1.3. This study

Although denial of trade-offs due to residents' protected values is one of the most significant issues regarding public acceptance, few researchers have investigated how to convince people with protected values to accept a public project. Hatori & Kajiwara (2012) showed that project acceptance by those with protected values depends on the associated decision-making processes. However, this finding was based on data obtained from university students, a somewhat restricted sample. Moreover, as the previous study only assessed the acceptability of decision procedures, little is known about the reasons why the acceptability of a proposal depends on the procedure used to make it. Accordingly, we replicated the previous study using an adult sample to determine which decision procedures can be used to moderate the negative attitudes of people with protected values. Furthermore, we investigated how people with protected values perceived the fairness of decision procedures, in a comparative manner, as this is known to contribute to public acceptance (Lind & Tyler, 1988).

4.2. Method

4.2.1. Participants

Three hundred residents of three cities (Makassar, Maros, and Watampone) in South Sulawesi were recruited randomly to participate in a questionnaire survey. The sample consisted of 148 males (49.3%) and 152 females (50.7%), and their mean age (standard deviation) was 33.46 (10.92) years (range of 19–65 years). Other information about personal attributes of participants is shown in Table 4.1.

	п	%
Respondence type		
Permanent house	229	76.3
Rental house	41	13.7
Others :	30	10.0
Occuaption		
Employee	46	15.3
Civil servants	119	39.7
Part time job	10	3.3
Student	65	21.7
House wife	21	7.0
Self employed	24	8.0
others :	15	5.0
Education		
<high graduate<="" school="" td=""><td>10</td><td>3.4</td></high>	10	3.4
High school	98	32.9
College degree	144	48.3
Graduate degree	46	15.4

Table 4. 1. Personal attributes of participants

4.2.2. Questionnaire and procedure

Questionnaires were administered by one member of a team of seven interviewers at the respondents' homes. Before undertaking the survey, the interviewers attended a 3-hour briefing session about how to administer the questionnaire and were informed about the objectives of the study. Each interviewer administered 35–45 questionnaires, and the average duration of the questionnaire was 30 min. Each respondent was informed that his or her responses would be used for research purposes only and was assured of confidentiality.

Measures of protected values. Initially, participants were asked to read a scenario about a dam construction project that may cause some fish species to become extinct. After reading the description, their tendencies to express protected values regarding the project were measured according to Baron & Spranca (1997). They were asked to select the option closest to their opinion regarding the project, from the following three options:

(1) The project should be prohibited no matter how great its benefit.

(2) The project should be accepted if it provides a sufficient benefit.

(3) I agree with the project.

As suggested by Baron & Spranca (1997), participants selecting choice 1 were identified as possessing protected values about this project. Hereafter, these are called PVs; those who selected choices 2 or 3 are called non-PVs.

The participants were then asked to rate their agreement with three items regarding the deontological rule using 7-point scales ranging from $1 = \text{totally disagree to 7} = \text{totally agree: moral obligation ("We have an obligation to try to stop this project"), anger ("I feel angry about this project"), and resentment ("I feel resentment towards this project"). To assess quantity insensitivity, the participants were asked whether it is equally wrong to allow this project to be implemented once or twice. Moreover, to measure omission bias, we asked the participants to read a new scenario in which more harmful consequences (extinction of more species) could result from not constructing this dam. After reading this scenario, they indicated their agreement with this project using 7-point scales ranging from 1 = totally disagree to 7 = totally agree.$

Decision procedures and acceptance. The participants were asked to read four scenarios in which this project was adopted based on the different methods shown in Table 4.2. After reading each scenario, the participants were asked to indicate their level of agreement with the decision using 7-point scales ranging from 1 = totally disagree to 7 = totally agree.

[Cost-benefit analysis]	
The government relied on cost benefit analysis (CBA) to make a decision about wh	ether
to build the dam. The economic cost of loss of fish species is included as a cost	
component of the project. The government estimated the cost of fish species extinc	tion •
be 100 million dollars in the case of the extinction of one fish species. Having comp	ared
the costs and benefits accrued from the dam project, the benefits from the project	were
found to be higher than the overall cost. Therefore, the government decided to impl	emen
the project.	
[Compliance with laws and customs]	
The government relied on traditional custom or related laws to make a decision abo	ut
whether to build the dam. The dam is to be located in an area that is approved by	
environmental law. Therefore, the government has decided to implement the projec	t.
[Majority rule]	
The government relied on the majority opinion of residents to make a decision abou	t
whether to build the dam. The government arranged a nationwide poll that included	ł
residents on both sides, for and against the project. The project was accepted base	ed on
the results of the nationwide opinion poll. Therefore, the government decided to	
implement the project.	
[Deliberation]	
The government relied on a deliberation process to decide whether to build the dam	ı. The
results of the discussion indicated support for the dam project. Therefore, the gove	rnme
decided to implement the project.	

Table 4. 2. Scenario for decision procedures

The fairness of each procedure was also measured by two questions: procedural fairness 1 ("This procedure is a fair way to implement the project") and procedural fairness 2 ("This procedure provides fair treatment to those involved"). Furthermore, we included three questions related to procedural fairness: dignity & respect ("This procedure treats residents with dignity and respect"), residents' rights ("This procedure respects resident's rights"), and anger ("How angry would you be about this outcome?"). All items were rated on a 7-point scale ranging from 1 = totally disagree to 7 = totally agree.

4.3. Results

4.3.1. Prevalence and properties of protected values

The proportion of people with protected values was 13.8%. The proportion of protected values did not vary significantly between males (11.0 %) and females (16.6 %). Also, no significant differences in age were found between participants with protected values (*Mean* = 33.83, *SD* = 11.10) and those without protected values (*Mean* = 33.37, *SD* = 10.95).

We compared participants with and without protected values with respect to the four characteristics of the deontological rule, and the results are shown in Table 4.3. Participants with protected values tended to possess more psychological traits associated with the deontological rule than participants without protected values (t = 3.03, p < 0.01 for moral obligation; t = 2.01, p < 0.05 for anger; t = 3.24, p < 0.01 for resentment; t = -3.13, p < 0.01 for omission bias). The response rate, which was insensitive to quantity, was higher from participants with protected values than from those without ($\chi^2 = 6.96$, p < 0.01 for quantity insensitivity). These results demonstrate the validity of this method for measuring protected value.

		-				
	Р	Vs	Non-	·PVs		
-	Μ	SD	Μ	SD	<i>t</i> -value	<i>p</i> -value
Moral obligation	368	2.22	2.72	183	3.03	0.003
Anger	332	208	259	260	201	0.048
Resentment	351	210	253	1.76	3.24	0.001
Omissionbiæs	4.63	193	5.48	1.54	-313	0.002
	%	n	%	n	x²	<i>p</i> -value
Quantity insensitivity	725	29	50.0	118	6.96	0.008

 Table 4. 3. Characteristics of deontological rules associated with protected values

4.3.2. Acceptance associated with protected values and decision rules

The means of acceptance of the four procedures, for participants with and without protected values, are shown in Figure 4.1. Participants with protected values tended to be less likely to accept a project that was adopted based on a cost-benefit analysis than another procedure. A project that was adopted according to laws and customs, rather than the other procedures, was less accepted by participants without protected values. The mean levels of acceptance for a project that was adopted through deliberation were highest for both groups.



Figures 4. 1. Acceptance associated with protected values and decision rules

Furthermore, we carried out a 2 (possession of protected values: with vs. without) × 4 (decision procedures: cost-benefit analysis, compliance with laws and customs, majority rule, and deliberation) repeated-measures analysis of variance, with acceptance as the dependent variable. The results revealed a significant interaction between possession of protected values and decision procedures (F(3, 294) = 4.32, p < .01). Our multiple comparison analysis revealed that participants with protected values were significantly less likely to accept cost-benefit analysis than majority rule and deliberation. We also found that the mean acceptance rates of cost-benefit analysis and compliance with laws and customs were significantly lower than for deliberation among participants without protected values.

4.3.3. Assessments of decision procedures

Table 4.4. shows the mean scores of assessments by protected values for the four decision procedures. Apart from the case of angry, significant differences between decision procedures were detected. According to the results of our multiple comparison analysis, the mean scores for cost-benefit analysis were significantly lower than those for majority rule and deliberation.
	Cost-	benefit	Complia	nce with	Major	it rulo	Dolih	orotion		
	ana	lysis	laws &	customs	IViajOr	ity rule	Dende	eracion		
	М	SD	Μ	SD	М	SD	М	SD	<i>F</i> -value	<i>p</i> -value
Procedural fairness_1	3.66	2.08	4.32	1.92	4.76	1.51	5.00	1.92	6.61	< 0.001
Procedural fairness_2	3.60	1.77	4.43	1.75	4.90	1.58	5.18	1.72	9.66	< 0.001
Dignity & respect	3.59	1.90	4.41	1.86	4.78	1.70	5.22	1.64	10.25	< 0.001
Residents' rights	4.00	1.99	4.80	1.68	5.05	1.75	5.24	1.85	6.06	0.001
Angry	3.76	1.96	3.41	1.53	3.34	1.46	3.10	1.86	1.66	0.18

Table 4. 4. Assessments of decision procedures for participants with protected values

4.4. Discussion

The rigid and absolute attitudes associated with protected values mean that public acceptance, in the context of regional policy making, may be impeded. However, the results of this study suggest that acceptability to those who hold protected values can vary depending on the procedures by which a decision is made. This shows that respondents with protected values become less likely to accept a project proposal if it is presented on the basis of cost-benefit analysis, as opposed to another method. These results are consistent with the findings of the previous study by Hatori & Kajiwara (2012). The fact that the same tendency was observed in different surveys involving data obtained from different samples (i.e., university students in Japan and adults in Indonesia) confirms the robustness of the findings. The new findings of the current study were that respondents' perceptions of fairness also vary, depending on the procedures, in the same manner.

Thus, the low acceptability of cost-benefit analysis in relation to respondents with protected values may be due to their negative rating of cost-benefit analysis on the basis of procedural fairness. Indeed, they were less likely to feel that this procedure was fair and treated people with dignity and respect. It may be that they thought their values should not be traded off in monetary terms. We found that project proposals developed through deliberation were more likely to be accepted by respondents with protected values. Their perception of procedural fairness was also rated higher. These results suggest that, even if people originally think that a project proposal conflicts with their protected

values, they might come to accept it as long the decision procedure is perceived to be fair and their values are taken into consideration.

These findings can be related to the value protection model developed by Skitka (2002). The model claims that people are motivated to protect their sense of personal identity when it is threatened, and they do so by making cognitive, affective, and behavioral adjustments. All of these impact whether they will feel that an event is fair or unfair. According to this model, protected values can have a negative effect on perceived procedural fairness. Less is known, however, about whether and how such an effect depends on the nature of the decision procedures, as few studies have considered different procedures. The results of this study suggest that the effects of protected values on perceived procedural fairness can be moderate, depending on the procedures.

It is important to note that the current research does not claim that the governing authority should not adopt the method of cost-benefit analysis in deciding whether to implement project proposals. Rather, we claim that authorities should pay attention to the possibility that such a method can incur strong opposition from people with protected values. Our findings imply that recruiting support from people who hold protected values requires their understanding that their values have been taken into consideration during the decision-making process.

Note that the present study was conducted based on a scenario survey with a hypothetical project. It is important to study residents' protected values in the context of actual projects, and the effects of decision procedures on their acceptability. Moreover, while this study stressed the effect of deliberation, less is known about how best to facilitate deliberation among people with protected values. This is another important issue to be addressed in future studies.

CHAPTER 5

Protected Values and Their Influences on Perceived Procedural Fairness in Highway Project: A Field Survey in South Sulawesi, Indonesia

5.1. Introduction

Public acceptance is key to the successful implementation of regional policies (Wan C, Shen Cg, Choi S., (2017). If the public broadly accepts a project proposal, governmental authority is able to smoothly implement it. Gaining support from residents may also help authorities raise funds for the project (Basbas S et.al., 2015). On the other hand, low levels of acceptance among residents may create obstacles to the progress of a project despite the best efforts to implement it (Payne et.al., 1996). As a result, a project may not be implemented in the face of strong public opposition.

Resident responses to government project proposals depend on perceptions of its benefits and costs and how each resident makes trade-offs between them (Hamersma et.al.,2016). Regional projects are commonly constrained by various trade-offs between incompatible values, as satisfying one value may entail sacrificing another. For example, highway projects can improve regional accessibility and reduce travel time, but they can also lead to changes in the local environment. Moreover, some projects displace neighbourhood residents. As long as residents' responses to proposed projects either directly or indirectly affect regional decision making regarding the project, residents need to understand such trade-offs to make well-reasoned judgments about the projects.

However, some people with strong values think that their values should not be traded off with other values. Many of these values concern human life, natural resources, and human rights. Baron and Spranca (1997) called values that are protected against trade-offs with other values *protected values*. Using terminology from economics, protected values are such that the marginal rate of substitution is infinite. People with protected values think that these values should not be sacrificed for anything, regardless of the benefit. In a regional decision-making process, such a refusal to make trade-offs by some stakeholders creates problems for government agencies, which try to allocate resources after

considering the values of the stakeholders. For instance, one stakeholder could dominate a decision by expressing an absolute value, or stakeholders with conflicting protected values could make a decision impossible (Baron & Leshner, 2000). Taken together, residents' denial of trade- offs due to loyalty to their protected values may challenge authorities' attempts to make reasoned judgments and garner public acceptance of decisions in diverse societies.

Although residents' denial of trade-offs due to loyalty to their protected values is one of the most significant issues regarding public acceptance, few studies have dealt with this issue in the context of regional public policy making. Furthermore, previous studies by Baron and his colleagues have explored protected values using hypothetical scenarios (Lim Cs, Baron J,1997). Little is known about whether people have protected values when it comes to actual projects, and, if they do, how such protected values affect acceptance of the projects.

Our study addressed the problematic characteristics of protected values that impede public acceptance and examined how to mitigate them. We examined a highway construction project in the South Sulawesi province of Indonesia to explore the nature and prevalence of protected values relevant to the project. This study attempted to demonstrate the effects of protected values on perceptions of procedural fairness, which is widely known to be an important determinant of public acceptance in various policy domains, including transportation policies (Gärling T, Jakobsson C, Loukopoulos P, Fujii S, 2008). We predicted and examined a detrimental effect of protected values based on the value protection model of justice (Skitka LJ., 2002). Finally, the study explored the role of trust in reducing the detrimental effects of protected values.

5.2.Theory

5.2.1. Properties of protected values

Baron and Spranca (1997) reported that a protected value is based on the *deontological rule*, which is contrasted with consequentialism and represents the normative position that judges the morality of an action based on its adherence to a rule or rules. It can be described as duty-, obligation-,

40

or rule-based beliefs because these beliefs stipulate that people conform to certain standards of behavior. Actions are more important than consequences for people who possess protected values. They have a belief in obligatory support or nonsupport for a certain action, regardless of the result of the action. Protectd values have four characteristics derived from the standpoint of the deontological rule: moral obligation, quantitative insensitivity, anger, and absoluteness (Baron and Spranca, 1997). First, the actions required or prohibited by protected values are regarded as moral obligations, as they are universal and objective, not simply conventions or personal preferences. Second, protected values make people insensitive to consequences. For example, an environmentalist with protected values related to environmental conservation considers destroying a species through a single act to be as bad as destroying a hundred species through a single act. Third, people may become angry if their protected values are violated, because they see it as a moral violation. Fourth, protected values can be associated with absoluteness, as people believe that these values should be protected from any trade-offs under any circumstances. In this study, we examined whether these properties could be observed among Indonesians with regard to their attitudes toward to a regional project.

5.2.2. Protected values and procedural fairness

Since the pioneering work of Thibaut and Walker (1975), which first presented the idea of procedural fairness, a large body of literature has demonstrated that perceptions of *procedural fairness* are important predictors of policy acceptance (Lind EA, Tyler TR, 1988, Tyler TR, Boekman RJ, Smith Hj, Huo YJ, 1997). The term procedural fairness is defined as the fairness of the procedures used to determine policy outcomes (Lind EA, Tyler TR, 1998). This is different from the concept of outcome fairness, which relates to the distribution of the costs and benefits within society (http://www.scirp.org). Much evidence suggesting that people are more likely to accept policy decisions that come from fair procedures than those from unfair procedures has accumulated within the field of social psychology (Tyler TR, Boekman RJ, Smith Hj, Huo YJ, 1997, Anand P, 2001,

Tyler TR, et.al.,1985). For example, Tyler et al.,(1985) found that the endorsement of the Reagan administration's tax policies in the United States was more strongly influenced by judgments of procedural fairness than by other outcome-related concerns.

The concept of procedural fairness is particularly important in light of efforts to promote public acceptance during regional decision-making processes involving trade-offs among values. Stakeholders usually have a variety of values that are more or less mutually incompatible or that even conflict with regional policy-making protocols. However, the observation that people also care about procedural fairness suggests that they can accept a regional policy if they perceive that it is fair, even if the policy is not attractive based on outcome-related concerns (Lind EA, Tyler TR, 1988).

Thus, a regional project tends to be accepted when the process surrounding the project is considered fair. However, people with protected values would likely reject the project even if they originally felt that the project procedure was fair. After recognizing that a project conflicts with their protected values, they might not feel that the procedure was/is fair. In particular, according to the *value protection model* developed by Skitka (2002), people are motivated to protect their sense of personal identity when it is threatened, and they do so by making cognitive, affective, and behavioral adjustments, all of which suggest whether they will feel an event is fair or unfair. According to this model, protected values can derogate from procedural fairness. Based on this background, the following hypothesis regarding the association between protected values and perceptions of procedural fairness was developed:

Hypothesis 1: Even if people originally feel the procedure for a regional project is fair, they may not think it is fair after recognizing that the project conflicts with their protected values.

The present study focused on the role of trust in government as a potential mitigator of the derogation effects of protected values on perceptions of procedural fairness. Fujii (2006) reported that trust in government affects people's perceptions of procedural fairness and their approval of a regional project. Even if people have protected values that conflict with a regional project, they may

believe that the procedure is fair and accept the project if they trust the government. This is particularly true in this case, as the study was investigating residents' attitudes towards a highway construction project not yet conducted. As a result, we expected that levels of trust in government would moderate the relationship between protected values and perceived procedural fairness. Accordingly, we proposed the following hypothesis.

Hypothesis 2: Trust in government mitigates the derogation effect of protected values on perceptions of the procedural fairness of the process surrounding a regional project.

5.3. Methods

5.3.1. Description of the flyover construction project in Indonesia

Our survey focused on the Simpang 5 flyover construction project that was being planned around three cities (Makassar, Maros, and Watampone) in the South Sulawesi province of Indonesia (Figure 5.1.). The intersection in front of the entrance to Hasanuddin International Airport at Makassar is a critical traffic convergence point for drivers travelling from Makassar to Maros and those traveling on the Reformasi toll highway that enters the airport. Drivers going from Watampone to Makassar or to the airport also use this intersection. In 2013, the government developed the Simpang 5 flyover and underpass construction project to connect Makassar with Maros (Figure 5.2.). The project cost RP. 300 billion/year (US \$30 million) over several years and was to be jointly funded by state and regional budgets. This project required 1.97 ha of land according to government regulation no. 71/2012, and the design called for a 1,050-m-long underpass that was 120 m long and 2×9 m wide.



.Figure 5. 1.Survey area



Figure 5. 2. The image of the Simpang 5 project

The government promised that completion of this project would prevent prolonged traffic gridlock, which is commonly seen around the intersection. However, a mosque (place of worship), residences, and public facilities, such as cemeteries, traditional markets, offices, and restaurants, around the intersection had to be moved or destroyed for this project. The government decided to follow the recommendations emerging from a cost–benefit analysis of the situation. To this end, a workshop was held with infrastructure experts and management companies to analyze the benefits and costs. The conclusion of the workshop was that the benefits were much higher than the costs. Accordingly, the government announced that it was proceeding with the project.

5.3.2. Respondents

Three hundred residents of three cities (Makassar, Maros, and Watampone; 100 respondents/city) were recruited randomly and participated in a questionnaire survey about the Simpang 5 flyover construction project. The sample consisted of 148 males (49.3%) and 152 females (50.7%), and their mean age (standard deviation) was 33.46 (10.92) years (range of 19-65 years).

5.4. Procedures and instruments

Questionnaires were administered by a one member of a team of seven interviewers at the homes of participants between March 12 and March 24, 2014. Before undertaking the survey, the interviewers attended a 3-hour briefing session about how to administer the questionnaire and were informed about the study objectives. Each interviewer administered 35–45 questionnaires, and the average duration/questionnaire was 30 min. Each participant was informed that his or her responses would be used for research purposes only and was assured of confidentiality.

In this survey, we assessed respondents' protected values against the Simpang 5 project. Their perceptions of the procedural fairness of cost-benefit analysis were measured twice; first, before they read the scenario that the Simpang 5 project adopted cost-benefit analysis to implement the project,

and a second time after they read the scenario. Using a within-subject design, the difference in each respondent's perception of procedural fairness between the first and second time was assessed to examine the influence of his or her protected values on the perceived procedural fairness for the project.

Pre-procedural fairness. Respondents were asked two questions related to the procedural fairness of the cost-benefit analysis. "This procedure (cost-benefit analysis) is a fair way to implement the project" and This procedure (cost-benefit-analysis) provides the fair treatment of those involved". Respondent indicated their agreement with statements using a 7-point Likert scale ranging from 1 = totally disagree to 7 = totally agree. These two ratings were added to yield a single score. The Cronbach's alpha of this measure was 0.74, which is high and indicates that the measure was reliable. This measure, which assessed perceptions of *pre-procedural fairness*, addressed each respondent's original feeling about the procedural fairness of the cost-benefit analysis (i.e., feelings before they were informed about the actual *Simpang 5* project procedure).

Attitude toward the project. Respondents were asked to read an explanation of the *Simpang* 5 project. After they read the description, their tendency to possess protected values regarding the project was measured according to Baron and Spranca (1997). They were asked to choose the opinion from the following three opinions that was the closest to their opinion regarding the project:

1. The project should be prohibited no matter how great its benefit.

- 2. The project should be accepted if it provides a sufficient benefit.
- 3. I agree with the project.

As suggested by Baron and Spranca (1997), respondents selecting choice 1 were identified as possessing protected values about this project. Hereafter, they are called *PVs*; those who selected choices 2 or 3 are called *non-PVs*.

Respondents were then asked to rate their agreement with four items regarding the deontological rule: absoluteness ("I cannot think of any benefit of allowing this project to proceed"),

moral obligation ("We have an obligation to try to stop this project"), anger ("I am angry about this project"), and degree of insensitivity ("It is equally wrong to allow this project to be implemented once or twice"). All items were rated on a 7-point scale ranging from 1=totally disagree to 7=totally agree. Additionally, trust in government was assessed by asking respondents to use a 7-point scale ranging from 1=not at all to 7=very much so to rate the extent to which they trusted the government.

Post-procedural fairness. Respondents were asked to rate the fairness of the procedure (costbenefit analysis) used for the *Simpang 5* project . We also created an additional two-item scale, which had an alpha relalibility of 0.80. This measure, which assessed perceptions of *post-procedural fairness*, addressed respondents' feeling about the procedural fainess of the cost-benefit analysis after they were informed about the actual Simpang 5 project procedure.

5.5. Results

5.5.1. Possession and properties of protected values.

The proportions of people with protected values were 22% in Makassar, 21% in Maros, and 26% in Bone. In terms of the reasons behind the protected values related to the Simpang 5 project, 51% of respondents cited religious reasons based on the fact that the project would destroy a mosque. Yet, 34% of respondents had protected values based on the fact that the project would force some residents to move.

To examine associations between a binary variable representing the possession of protected values and demographic variables, odds ratios (ORs) and 95% confidence intervals (CIs) for each level of each variable were calculated. As shown in Table 5.1., females were more likely to possess protected values than were males. It was also shown that people who were 45-54 years of age, had been public servants, or who had graduate degrees were less likely to possess protected values.

Variable	OR	95% CI	P Value
Gender		1	
Male	-	-	-
Female	1.73	1.00-2.99	0.05
Age			
18-24	-	-	-
25-34	0.81	0.41-1.59	0.54
35-44	0.64	0.32-1.31	0.22
45-54	0.27	0.10-0.77	0.01
55≤	0	0.00-0.00	1
Occupation			
Employee	-	-	-
Public servant	0.41	0.18-0.91	0.03
Part time job	1.52	0.37-6.26	0.56
Student	1.17	0.52-2.63	0.71
Housewife	0.25	0.05-1.25	0.09
Self-employed	0.46	0.13-1.59	0.22
Others	1.14	0.33-3.96	0.83
Education		1	
<high graduate<="" school="" td=""><td>-</td><td>-</td><td>-</td></high>	-	-	-
High school	0.61	0.17-2.22	0.46
College degree	0.72	0.21-2.54	0.61
Graduate degree	0.19	0.04-0.92	0.04

Table 5.1. Odds ratios for possession of protected values (n=300).

5.5.2. Properties of protected values

The four characteristics of the deontological rule were compared between PVs and non-PVs, and these results are shown in Figure 5.3. PVs tended to possess more psychological traits associated with the deontological rule that did non-PVs (t = -1.78, p = 0.07 for absoluteness; t = -4.86, p = 0.00

for moral obligation; t = -5.36, p = 0.00 for anger; and t = -3.31, p = 0.00 for quantity insensitivity). These results demonstrate the validity of the current measure of protected values.



Figure 5. 3. Means of the properties of the deontological rule among people with (PVs) and without (non-PVs) protected values

5.5.3. Impacts of protected values on procedural fairness

Figure 5.4. shows the differences between the pre-procedural and post-procedural fairness ratings for the PV and non-PV groups. PVs viewed the procedure of cost-benefit analysis to be significantly less fair at the end than they did initially (t = 2.74, p = 0.01), whereas non-PVs had similar reactions to pre- and post-procedural fairness (t = 1.36, p = n.s.). This result indicates that respondents holding protected values did not think the project procedure was fair if they recognized that it was inconsistent with their values. Thus, perceptions of procedural fairness decreased in those with protected values. This result supports our first hypothesis .



Figure 5. 4. Means of perceptions of pre- and post-procedural fairness among people with (PVs) and without (non-PVs) protected values

5.5.4. Effect of trust in government

We examined whether the effect of protected values on perceptions of procedural fairness was mitigated by trust in the government. We classified respondents into higher and lower trust groups according to their score on the trust measure: respondents with scores in the top 50% of the sample were placed into the high-trust group, and those with scores in the bottom 50% of the sample were placed into the low-trust group. Then, we compared the means for perceptions of post-procedural fairness among the four groups (i.e., PVs and non-PVs in the high-trust groups and PVs and non-PVs in the low-trust groups). As shown in Table 5.2, the difference in the perceptions of post-procedural fairness held by PVs and non-PVs was small in the high-trust group, (t = 0.85, p = n.s.), whereas the difference was significant in the low-trust group (t = 3.58, p = 0.00). This result supports hypothesis 2.

 Table 5. 2. Means of post-perceptions of procedural fairness associated with protected values and trust in government

	PVs	Non-PVs
High Trust	5.08	5.38
Lower Trust	3.95	4.69

5.5.5. Structural equation model analysis

Finally, we estimated a structural equation model (SEM) to verify the causal relationship between procedural fairness and protected values. Trust was also included in this model as a mediator to reduce the detrimental effect of protected values on perceptions of procedural fairness. Figure 5.5. shows the estimated coefficients of the model. The model includes only paths that were significant at the 5% level. The link between trust and post-procedural fairness was omitted because it was not significant. The model fit was found to be acceptable: GFI=0.98, adjusted GFI=0.95, CFI=0.98, and RMSEA=0.06.



Figure 5. 5 Results of the SEM analysis for post=procedural analysis

Shown in Figure 5.5., protected values had a negative effect on post-procedural fairness. Preprocedural fairness was positively related to post-procedural fairness, and trust had a positive effect on pre-procedural fairness. Accordingly, trust indirectly affects post- procedural fairness through its influence on pre-procedural fairness. The total effects of these variables as the sums of direct and indirect effects on post-procedural fairness are shown in Table 5.3. Whereas post- procedural fairness was influenced most strongly by initial ratings of procedural fairness, the negative effects of protected values on post-procedural fairness as well as the mediating effect by trust were also found to be significant.

	Total effect	Direct effect	Indirect effect
Post-procedural fairness	0.59	0.59	0.00
PV	-0.22	-0.22	0.00
Trust	0.19	0.05	0.14

 Table 5. 3. Total effects (standardized of coeficients) on post-perception of procedural fairness

5.6. Discussion

The present survey showed that about 20% of respondents had protected values in opposition to the *Simpang 5* project. The proportion of protected responses was generally lower than that shown in previous studies examining the prevalence of protected values in a hypothetical choice. For example, the results of a survey by Lim and Baron (1997) showed that the mean proportion of respondents with protected values among 17 hypothetical choices was 57.24% (SD = 17.22) in a Malaysian sample, 59.06% (SD = 20.67) in a Singaporean sample, and 57.47% (SD = 14.18) in a US sample. A particularly relevant factor affecting the difference between actual decisions and hypothetical decisions is the importance of consequences (Kühberger et al., 2002). As discussed previously, protected values are based on a deontological rule that binds people to a certain decision, independent of the decision's consequences. Protected responses based on such a rule are more likely to appear in hypothetical decision scenarios when respondents are asked to make a hypothetical decision with hypothetical outcomes irrelevant to actual outcomes. On the other hand, as shown in this study, protected responses were less likely to appear in actual decision scenarios, as respondents tended more to take into account actual outcomes if they had to live with them. Yet, as demonstrated by Baron and Spranca (1997), our results show that some respondents also had protected values associated with the four properties of the deontological rule when making decisions related to an actual problem.

As already explained, previous studies have emphasized the importance of procedural fairness for residents' acceptance of a public decision (e.g., Lind & Tyler, 1988). Yet, the present results suggest that residents' perceptions of procedures fairness can be affected by the presence of protected values. It should be noted that residents' initial perceptions of procedural fairness were shown to still exert a significant impact on their views of the procedures even after they recognize that the project conflicted with their protected values. Even so, however, the present findings regarding the negative effect of protected values indicate that residents with protected values might oppose a project if their values were threatened by the project, even if the project followed fair procedures.

Another findings of our study support the role of trust in government as a mitigator of the derogation effect of protected values on perceptions of procedural fairness. As long as residents trust government, perceptions of procedural fairness can be maintained even if some residents have protected values that are relevant to regional projects. The government should make efforts to promote trusting relationships with residents through, for example, showing their sincerity. Our results, however, show that trust in government does not have a direct effect on the final perceptions of procedural fairness (post- procedural fairness). Furthermore, the negative correlation between trust and protected values suggests that people with protected values tend not to trust government. Thus, the ability of trust to mitigate the derogation effect of protected values may be limited. Therefore, even if trusting relationships between residents and the government can be formed, it would be difficult for the government to completely avoid a situation in which residents' perceptions of procedural fairness are diminished by protected values.

More generally, our findings suggest the need for more fundamental measures that focus on the nature of protected values and allow decision-makers to reach reasonable decisions about regional projects. As suggested by Baron and Leshner (2000), protected values result from unreflective overgeneralizations that lead to incorrect or overgeneralized concepts. People tend to develop protected values without giving sufficient thought to the possible benefits and costs related to the actions required or prohibited by their values. However, people may change even extremely strongly held attitudes by reflecting on their validity. It is important for local governments to communicate effectively with stakeholders so that those with protected values consider the implications of their opinions on the region.

Note that the present sample consisted predominantly of Muslims, a group that tends to have protected values that often lead to opposition to various projects based on religion. Although religion is likely to lead people to protect their own values (e.g., Lim & Baron, 1997), the relationship between protected values and religion has not been sufficiently examined theoretically or empirically. A cross-cultural study examining the prevalence of protected values in different cultures would contribute to understanding the effects of religion and other demographic variables, such as race and educational level, on the development of protected values. Additionally, given the socially harmful impact of protected values, it is also important to study the psychological processes by which protected values are developed or mitigated. Experimental research that examines the effects of reflective thinking and communication processes on protected values would contribute to developing relevant measures to reduce the negative impacts of protected values.

CHAPTER 6

Protected Values and Illusion of Understanding Around Risk Acceptance: A Vignette Experiment of the Construct of High Level Radioactive Waste

6.1. Introduction

The issue of forming consensus about taking significant community risks, such as the construction of high-level radioactive waste (hereinafter HLW) disposal facilities, incorporates a trade-off between compensation (benefits) when the risk is accepted and values impaired by doing so (Fischhoff and Kadvany, 2011). In such cases, conflict of interest between concerned parties with different interests and values may become marked; the pursuit of one set of values can lead to the impairment of another set of values. In particular, it becomes extremely difficult to form consensus regarding risk acceptance when the parties concerned insist that their values and opinions are absolutely non-negotiable. There are concerns that social decision-making may come to a stalemate in cases where people become polarized regarding a specific policy. This phenomenon is generally known as group polarization (Sunstein, 2000).

6.1.1. The concept of protected values

The abovementioned issues are related to protected values, which have been discussed by Baron and Spranca (1997). Protected values are generally defined as a set of values protected from trade-offs with other values, or an attitude where one will not compare or balance particular important values with other values. In other words, these values are based on an absolute belief that one's values must be protected at all costs. It has been pointed out that people may assign protected values to human and animal life, the natural environment, human rights, divinity, etc. So as not to condone any activities of development that could bring about extinction of plants/animals or acts of buying or selling organs for profit (Baron, 2008; Baron and Spranca, 1997). In addition, it has been pointed out that protected values that refuse the comparison of risks to benefits may be involved in the issue of

restarting nuclear power plants (Sogabe and Hatori, 2013) and the issue of the construction of HLW disposal facilities (Oonuma et al., 2015).

Unlike consequentialism, protected values are considered to be based on deontological rules regarding behavior itself rather than the consequence of the behavior (Baron and Spranca, 1997). Here, deontological rules generally represent rules that govern actions that should or should not be taken, regardless of their consequences. According to Baron and Spranca (1997), protected values may show qualities of 1) "quantity insensitivity," 2) "obligation," 3) "anger," and 4) "posturing" as they are based on such deontological rules (Baron and Spranca, 1997). Firstly, preserved values are not dependent on the consequence of an action and, therefore, one's attitude towards that action is quantitatively insensitive to the "quantity" of the consequence. For instance, those who hold protected values regarding forests tend to think that logging activity is wrong regardless of whether 100 hectares of forest are logged or 200 hectares are logged. Secondly, those holding specific protected values feel that executing that action or stopping that action is a duty imposed on themselves. This kind of sense of duty is considered to be based on an attitude where one tries to protect the values that he/she regards as absolute from trade-offs. Thirdly, those holding protected values tend to harbor a strong sense of anger towards any impairment of those protected values. With regard to this, Tetlock et al. (2000) pointed out the possibility that the very act of trading off one's values with other people's values may induce a sense of anger. Finally, those holding protected values tend to actively assert their opinions during discussions about said values.

When such protected values are involved in situations where decision making regarding public policies is required, it becomes difficult to compare/balance with other values; in some cases, only a specific set of values will be regarded as important, leading to the possibility that appropriate decision-making is not carried out (Baron and Leshner, 2000). In the case of the construction of HLW disposal facilities, Onuma et al. (2015) pointed out that the issue of protected values is unavoidable

when considering the consensus formation required for constructive discussion between those for and against the construction.

6.1.2. Protected values and the illusion of understanding

On the other hand, it is possible that protected values are not formed with sufficient consideration given to the significance or consequence that those values have (Baron and Leshner, 2000; Tetlock, et al., 2000). Based on this possibility, Baron and Leshner (2000) demonstrated that protected values may be attenuated in those holding them when they imagine counterexamples to those values (circumstances in which they had no choice but to accept the action they object to). Similarly, Hatori and Kajiwara (2014) used a dam project as the subject of their study and found that protected values may not be retained when people are given the opportunity to reflect on situations in which their protected values conflict with other values. As seen with protected values, these findings suggest that even if one regards a specific set of values or opinions as absolute, such attitudes can be transformed.

Thus, it is possible that those holding protected values may have not given enough thought to the significance and validity of those values. Now, when we think about those holding protected values who nonetheless regard their values as absolute, the possibility arises that they may have an erroneous perception that they "fully understand" the validity of those values. With respect to this lack of recognition of one's degree of understanding, studies on the "illusion of understanding" have been accumulated in the field of cognitive psychology (Keil, 2003). Rozenblit and Keil (2002) experimentally demonstrated that the general public is convinced that it has a good understanding of day-to-day matters and showed empirical findings about the illusion of understanding. In this experiment, the subjects were asked to describe the causal mechanism by which apparatuses such as helicopters or pianos operate. Subjects' self-assessment of understanding decreased after the experiment compared to before the experiment. This shows that people tend to falsely believe that they have a good understanding of day-to-day matters and demonstrates the existence of the illusion of understanding. In the present study, we followed the experimental procedures described above and examined the relationship between protected values and the illusion of understanding regarding risk acceptance.

6.1.3. Hypothesis of the present study

Based on the above discussion, we put forward the following hypothesis regarding the impact of the opportunity to independently reflect on one's social values (pros and cons) on self-assessment of understanding and holding protected values in a situation where risk acceptance is involved in the implementation of public policy.

Protected values regarding risk acceptance tend to be attenuated with a decrease in self-assessment of one's own understanding of the public policy in question through independent reflection on the social values (pros and cons) of the public policy.

If the above hypothesis is correct, it can be considered that those holding protected values regarding public policies have overestimated their understanding of these policies. At the same time, it is considered that this kind of illusion of understanding decreases through independently thinking about the social values of public policies, and protected values then tend to become attenuated.

In the present study, we looked at the construction of HLW disposal facilities as a consensusforming issue regarding risk acceptance in a community in order to verify the above hypothesis. Although there are Japanese laws governing the disposal of HLW, candidate sites for final disposal are yet to be determined and specific project plans remain to be drawn. In view of this, we conducted a vignette experiment survey with the construction of HLW disposal facilities as its theme and residence city as the subjects.

In the present experiment, we set up a reflection task where the subjects considered the pros and cons of constructing HLW disposal facilities; additionally, we asked them about their tendency to hold protected values regarding the construction of HLW disposal facilities and obtained their selfassessments of understanding before and after the task. Specifically, as a reflection task regarding HLW disposal facilities, we asked the subjects to freely describe the social values (pros and cons) of HLW disposal facilities. Here, according to Keil (2003), the illusion of understanding regarding matters of the real world is associated with the inability to adequately explain those matters. This being the case, it was expected that the self-assessments of participants' understanding would decrease if, through the reflection task, the participants realized that they were unable to adequately explain the social values (pros and cons) of HLW disposal facilities. Also, if the hypothesis of the present study was correct, we believed that the tendency to hold protected values would become attenuated through the task.

As mentioned above, studies conducted by Baron and Leshner (2000) and Hatori and Kajio (2014) have also confirmed that protected values tend to become attenuated when one independently reflects on counterexamples and conflict situations. However, these studies did not examine how people holding protected values evaluate their understanding of issues in question that require decision-making. Therefore, the relationship between protected values and the illusion of understanding remains unclear.

6.2. Experiment

6.2.1. Experimental participants

We conducted an experiment involving 102 students of Ehime University which using questionnaires. The participants included 67 men (65.7%) and 35 women (34.3%), with an average age of 18.89 years (standard deviation = 0.90 years).

6.2.2. Experimental procedures

The experimental procedures of the present study are shown in Figure 6.1.





First survey. We presented the participants, who were assumed to be residents of City A, with a virtual scenario where construction of an HLW disposal facility in A City was planned in order to assess the tendency to hold protected values regarding the construction. We asked the participants to select one choice that reflected their attitude toward the project: (1) "construction of a final disposal facility is absolutely unacceptable, no matter how much benefit it may bring"; (2) "construction of a final disposal facility is acceptable if it will bring enough benefits"; or (3) "I support the construction of a final disposal facility." Here, according to the classification presented by Baron and Spranca (1997), the participants that selected choice (1) were classified as protected value holders ("PV

group"), and the remaining participants were classified as non-protected value holders ("non-PV group").

We asked the participants to assess their understanding of the impact of HLW disposal facility construction and to respond on a 7-point scale ranging from "do not understand at all" to "understand well enough." We then created a scale on which "7" represented "understand well enough" and "1" represented "do not understand at all." Furthermore, in order to examine the degree of confidence held by the experiment participants with regard to the HLW disposal facility construction project, we created a scale for "confidence" and asked them to assess their opinions about the project on a 7-point scale ranging from "1: it may be wrong" to "7: it is not wrong." In addition to these questions, we asked the participants to respond to questions shown in Table 6.1. regarding the psychological characteristics of protected values, as mentioned earlier.

Table 6. 1. Measurements of the psycho	ological	characteristi	cs of	protected	value
and evaluations according to	o posses	ssion of prote	ected v	values	

	PV group	Non-PV group	
	Mean (SD)	Mean (SD)	t-value
1) Quantitative insensitivity			
Regardless of the risk level of radiation leakage, the construction of a final disposal facility is equally wrong.	3.96 (1.41)	3.00 (1.14)	3.78***
2) Sense of duty			
You think that "we are obligated to stop the construction of the final disposal facility."	4.76 (1.70)	3.50 (1.54)	3.93***
You think that the only moral choice for you is to call off the construction of the final disposal facility.	4.22 (1.19)	2.84 (1.26)	5.63***
3) Anger			
You feel resentment toward the construction of the final disposal facility.	4.04 (1.49)	2.73 (1.30)	4.75***
You become irritated just by thinking about the construction of the final disposal facility.	3.09 (1.75)	2.29 (1.35)	2.62**
4) Posturing			
(Please imagine that you are taking part in a civic debate about the construction of the final disposal facility)You think that you should actively express your opinion about the project in such situations.	5.70 (1.19)	4.73 (1.34)	3.80***

(Response on a 7-point scale ranging from "(7) strongly agree" to "(1) strongly disagree")

*** *p* < .001, ** *p* < .01

Second survey Two weeks after the first survey, we asked the participants to freely describe the pros

and cons of HLW disposal facility construction as a reflection task.

We then asked them to self-assess their tendency to hold protected values and their understanding of the HLW disposal facility construction project, as was done in the first survey. Subsequently, we again divided the participants into the "PV group" and "non-PV group" based on their responses.

6.3. Results

6.3.1. Tendency to hold protected values and associated psychological characteristics

Of the 102 experiment participants, 46 of them (45.1%) were identified as holding protected values and grouped into the PV group, while the remaining 56 (54.9%) were grouped into the non-PV group at the time of the first survey.

We then compared the related measures associated with the psychological characteristics of protected values between the PV group and non-PV group at the time of the first survey (Table 6.1.). The results showed that compared to the non-PV group, the PV group tended to show higher quantitative non-sensitivity (t = 3.78, p = .00), sense of duty (t = 3.93, p = .00; t = 5.63, p = .00), anger and resentment (t = 4.75, p = .00; t = 2.62, p < .01), and tendency to display their opinions (t = 3.80, p = .00). These results are consistent with what Baron and Spranca (1997) pointed out as the characteristics of protected values, and they once again suggest that protected values are based on deontological rules.

6.3.2. Changes in protected values and assessment of understanding

We compared the distribution of protected values before and after (at the time of the first and second survey) the reflection task regarding the social values of HLW disposal facilities (Table 6.2.). The results showed that 17 (37.0%) out of 46 participants in the PV group at the time of the first survey had converted to the non-PV group after completing the reflection task. On the other hand, 3 (5.4%) out of 56 participants in the non-PV group at the time of the first survey had converted to the PV group at the time of the first survey had converted to the PV group at the time of the first and second surveys showed that there

was a significant difference (45.1% at the time of the first survey, 31.4% at the time of the second survey, z = 2.01, p < .05).

	_	At the time of the second survey			
		PV group	Non-PV group	Total	
At the time	PV group	29 (63.0%)	17 (37.0%)	46 (100%)	
of the first survey	Non-PV group	3 (5.4%)	53 (94.6%)	56 (100%)	
	Total	32	70	102	

Table 6. 2. Distribution of the possession of protected values in two surveys

Furthermore, we compared the means of understanding and confidence regarding HLW disposal facilities before and after the reflection task (Table 6.3.). With respect to the understanding assessment, there was a significant trend in the difference between the first survey and the second survey, with the means tending to be higher at the time of the second survey. The confidence assessment showed no significant difference.

Table 6. 3. Evaluations of self-assessments of understanding in two surveys

	First survey	Second survey	
	Mean (SD)	Mean (SD)	t value
Understanding assessment	3.47 (1.45)	3.72 (1.26)	-1.71 †
Confidence assessment	3.16 (1.64)	3.08 (1.49)	.50

p < .10

6.3.3. Relationship between protected values and assessment of understanding

We examined how the understanding and confidence assessments regarding HLW disposal facilities changed among those whose protected values became attenuated and those whose protected values did not become attenuated through the reflection task. We grouped those who held protected

values at the time of the first survey (PV group) and those who converted from the PV group to the non-PV group at the time of the second survey into the "PV conversion group," grouping the rest into the "PV non-conversion group"; we then compared the between-group difference in the understanding assessment and confidence assessment between the first and second surveys. The results are shown in Figure 6.2. and Figure 6.3. As the figures show, both understanding and confidence decreased in the PV conversion group, while there was a trend of increase in the PV non-conversion group. We therefore performed a two-factor repeated measures ANOVA on the survey time points (the first and second surveys) and whether PV conversion took place (the PV conversion group and the PV non-conversion group). The results showed that there was a significant interaction between the two factors in the understanding assessment (F(1, 44) = 9.73, p < .01), and there was a significant trend in the confidence assessment (F(1, 44) = 3.13, p = .08).



XValues in brackets represent standard deviation

Figure 6. 2. Means of self-assessments of understanding before and after the experimental work associated with the change in protected values



XValues in brackets represent standard deviation

Figure 6. 3. Means of self-assessments of confidence before and after the experimental work associated with the change in protected values

6.4. Discussion

The present experiment showed that the possession of protected values regarding the HLW disposal facility construction project tended to decrease through the reflection task addressing the social values of the project. In particular, it was shown that in approximately 40% of those who held protected values, their protected values tended to be attenuated through the experiment. Furthermore, there was a relationship between the attenuating tendency of protected values and the decrease in the understanding assessment before and after the experiment task. Those whose self-assessments of understanding the HLW disposal facilities decreased tended to show attenuation in protected values. From a different perspective, these results indicate that those who held protected values regarding the HLW disposal facility construction project prior to the experiment assessed their own understanding of the project as high, and it is highly possible that they had developed the illusion of understanding discussed by Rozenblit and Keil (2002). We therefore believe that protected values became attenuated

as the illusion of understanding became attenuated through the explanation task of the present experiment. The above results support the hypothesis of the present study.

The above discussion suggests that those who held protected values regarding the HLW disposal facility construction project may face a double challenge in properly assessing the social values of the project. First, it is possible that those holding protected values may not give enough thought to the social values (pros and cons) of HLW disposal facilities in the first place. This possibility is consistent with the findings of previous studies (Baron and Leshner, 2000; Hatori, Kajio, 2014) indicating that those holding protected values do not give enough thought to situations where the values they believe in come into conflict with other values. Second, it is possible that, despite not having given enough thought to the decision in question, as mentioned earlier, those holding protected values may falsely believe that they have a good understanding. Therefore, there is concern that such people may hold an illusion of understanding where they are unaware that they do not understand.

6.4.1. Implications for the issue of consensus formation

Here, we discuss the results of the present study in terms of consensus formation regarding risk acceptance, which arises in relation to public policies and their implementation.

First, the results of the present study suggest that protected values regarding public policies and risk acceptance may not necessarily be absolute or invariable; they may be altered, depending on the situation. In other words, it is possible that those holding protected values may not give enough "thought" to their values and, therefore, those protected values can be expected to spontaneously transform if the individual thinks about the validity of those protected values. In situations where consensus formation on public policies is required, it is not uncommon for the decision-making process to be affected by "those with a strong opinion" and "those that are vocal" (c.f. Kuwako, 2011). However, the results of the present study imply that the opinions of "those with a strong opinion" and "those that are vocal" may not necessarily have been formed through an adequate thinking process. Decision makers in charge of public policies should pay careful attention to this point and encourage consensus formation among stakeholders.

Second, based on the results of the present study, it is important to provide opportunities for independent reflection where stakeholders can think about the social values of the policies in question in order to promote consensus formation on public policies. The issue of whether people will accept a public policy is generally called a public acceptance issue, and various studies on the topic have been conducted so far (Aoki, 2006; Fujii, 2003). With regard to the HLW disposal facility construction issue, the importance of procedural justice and trust (Ootomo et al., 2014; Sakamoto and Kanda, 2002) has been pointed out as a condition for promoting public acceptance, and this has provided important suggestions for considering the decision-making process and consensusformation process regarding public policies. However, previous studies have not adequately or empirically examined what effects a reflection process (in which one "thinks" about the social values of public policies) has. The results of the present study suggest that such reflection opportunities play an important role in attenuating people's attitudes that render their opinions and values absolute. As mentioned in the beginning, conflict of interest regarding risk acceptance may become serious when the stakeholders adopt a tough attitude, as seen in protected values. It is important to facilitate communication that promotes reflection on the policies and attenuates the problem of protected values in order to mitigate such conflicts of interest.

Now, a specific approach to providing such reflection opportunities may be to arrange an opportunity to think about the pros and cons of the policy, as was done in the present study, for instance when facilitating civic dialog or risk communication between experts and the public regarding public policies. As mentioned earlier, the effects of such reflection opportunities may include people becoming aware that their understanding of the policies in question is not complete, in addition to thinking about both the pros and cons of the public policies. It is expected that these effects will promote the attenuation of protected values regarding the policies and encourage sounder

judgment in participants. In particular, the approach used in the present study was designed with the intention that the participants would think for themselves about the social values of public policies, and it is unique in that its purpose was not to sway the participants to follow a particular set of values or opinions. In this regard, the approach of the present study eliminated arbitrary guidance intended to sway participants to a specific opinion as much as possible; it was designed for each participant to spontaneously come to an appropriate social decision about the policy in question.

6.4.2. Subjects for future investigation

Finally, we will discuss subjects for future investigation. First, a virtual scenario regarding an HLW disposal facility construction project was presented to residence of Makassar, and the protected values of the parties involved in actual HLW disposal facility construction projects were not examined. At present, candidate HLW disposal sites have yet to be determined in Makassar, Indonesia. However, there are a number of municipalities where the issue of inviting HLW disposal facilities has arisen (Uemura and Kawamoto, 2008). Important issues to be addressed in the future include focusing on such municipalities and shedding light on both the protected values held by local residents and the challenges within their individual and specific contexts. Secondly, although the present study showed that there was an interrelationship between the possession of protected values and the illusion of understanding regarding risk acceptance, the psychological mechanism behind the relationship and causal association remains to be elucidated. Here, the illusion of understanding, which is a psychological phenomenon, is a problem associated with the metacognition of one's own understanding (Everson and Tobias, 1998; Sannomiya, 2008). We may be able to gain further insight into the relationship between protected values and the illusion of understanding by examining the cognitive and learning processes regarding protected values from the perspective of metacognition. Finally, a variety of specific approaches are conceivable for introducing reflection opportunities in settings of consensus-formation processes and risk communication regarding public policies. We will

further examine how to arrange reflection opportunities in order to encourage the attenuation of protected values in real-life situations of consensus forming and risk communication. Verification of the effects through social experiments is also an important task.

CHAPTER 7 General Conclusions

This section includes the conclusions of the study, the new findings and the implications of the literature and/or policies.

The process of making decisions is a method in solving a problem which considers various factors as well as the desired outcome one wishes to achieve. A good decision is obtained when all stakeholders were accepting of it without any conditions. In many developed countries, the infrastructure projects carried out by the government generally do not get good levels of acceptance from the residents. For instance, the highway construction project, waste disposal facilities, and displacement policy. Mainly, projects that have a direct impact on the resident such as removing or demolishing a sacred place, relocating residents to a new location and environmental damage, the residents will have a strong opinion or protected value in opposition to the projects. This study focused on the psychological aspects as relevant factors that have effects on residents' acceptance of actual projects, to which minimal attention has been given by previous research.

This research has attempted to explore a deeper understanding of the residents' attitudes toward infrastructure projects in Indonesia. Based on the research findings, we conclude that protected values have correlation with consensus building (acceptance), when protected values are higher, then acceptance is lower and vice-versa. For instance, when the public disagreed with the project proposal, the government was unable to implement or even to postpone it. However, if the public agreed with the project, the government was able to implement the project smoothly. Gaining support from the residents may also assist the authorities in raising project funds. It is proven that protected values have correlation with consensus building (acceptance).

Furthermore, the study also found that protected values have association with procedural fairness in the decision process of infrastructure projects. In this case, perceived distributional fairness of costs, environmental burdens and procedural fairness are linked to a higher public acceptability

70

and can decrease protected values. We found that project proposals developed through deliberation were more likely to be accepted by respondents with protected values. Their perception of procedural fairness was also rated higher. These results suggest that, even if people originally think that a project proposal conflicts with their protected values, they might come to accept it as long as the decision procedure is fair and valuable. We may conclude that people's protected values can be overcome or reduced by fair procedures. As it is known, the higher the protected value, the smaller the acceptance; but if the project is implemented with fair procedures, the high protected value can decrease, and then public acceptance will increase as well.

This study seeks also to develop an effective policy measure to mitigate negative impacts of protected values. The decision-makers who are in charge of public policies should pay attention on this point and realize that they cannot gain residents' acceptance of infrastructure projects simply by offering compensation. They should consider the importance of residents' attitude, behavior and their trust to implement this project. This assessment will encourage consensus formation among stakeholders. It is important to provide opportunities for independent reflection where stakeholders can consider the social values of the policies in question in order to promote consensus formation on public policies. A reflection opportunity plays an important role in attenuating people's attitudes that render not only their opinions and values, but also provides an opportunity to think about the pros and cons of the policy. Last but not least, a communication facility that promotes reflection on the policies and attenuates the problem of protected values in order to mitigate such negative impacts of protected values.

We suggest that public involvement, consultation and independent reflection should become the important things to consider in decision policy-making. This might be an effective strategy to introduce the project by providing information about the processes and effect of the project to residents, creating the exchange of adequate information flow between government and resident. This may serve as a cornerstone for the project process, as it found trust is related between government and residents, and it provides the opportunity for resident's concerns to be counted. Finally, this is an essential of policy measure to encourage the planning and implementation of infrastructure projects.
References

- 青木俊明 (2006). 胆沢ダム建設に対する一般市民の賛否態度の形成構造-公正理論を用いた実 証分析-,都市計画学論文集,41,761-766.
- Asnawati M. (2019). The modernization impacts for Indonesian development. *Asian Journal of Social Sciences & Humanities Vol. 8*(1).
- Antara News. (2012). Siap-siap, nanti air datang (in Bahasa). Retrieved November 1, 2016, Retrieved from http://www.antaranews.com/print/293148/siap-siap-nanti-air-datang.
- Antlöv, H. & Cederroth, S., (1994). Leadership on Java: Gentle Hints, Authoritarian Rule. London: Marston Book Services.
- Baron, J. (2008). Thinking and Deciding (4th ed). Cambridge University Press.
- Baron, J. & Spranca, M. (1997). Protected values. Organizational Behavior and Human Decision Processes, 70, 1–16.
- Baron, J., & Leshner, S. (2000). How serious are expressions of protected values. *Journal of Experimental Psychology*, 6, 183-194.
- Basbas, S., Mintsis, G., Taxiltaris, C., Roukouni, A., & Vazakidis, A. (2015). Public acceptance in financing urban transportation projects using the Value Capture concept. In C. A. Brebbia & J. L. Miralles i Garcia (Eds.), Urban Transport XXI (pp. 613-625). Southampton: WIT Press.
- Bronfman, N. C., Jiménez, R. B., Arévalo, P. C., & Cifuentes, L. A. (2012). Understanding social acceptance of electricity generation sources. *Energy Policy*, 46, 246-252. https://doi.org/10.1016/j.enpol.2012.03.057
- Cropanzano, R., & Greenberg, J. (1997). Progress in organizational justice: Tunneling through the maze. International Review of Industrial and Organizational Psychology, 12, 317–372.
- De Groot, J.I.M.; Schuitema, G. (2012). How to make the unpopular popular? Policy characteristics, social norms and the acceptability of environmental policies. *Environ. Sci. Policy*, *19–20*, 100–107.
- Dirar, A., Moghraby, A., & Hashim, M. J. (2015). *Displacement and resistance induced by the Merowe Dam: the influence of international norms and justice*. United Kingdom.
- Dreyer, S.J. & Walker, I. (2013). Acceptance and Support of the Australian Carbon Policy, *Soc Just Res 26*: 343. https://doi.org/10.1007/s11211-013-0191-1.
- Drews, S.; van den Bergh, J.C.J.M. (2015). What explains public support for climate policies? A review of empirical and experimental studies. *Clim. Policy*.
- Dorcey, A., Doney, L. and Reuggeberg, H., (1994). Public Involvement in Government Decision-Making: Choosing the Right Model, B.C. Round Table on the Environment and the Economy, Victoria.
- Dwyer, W.O., Leeming, F.C., Cobern, M.K., Porter, B.E., & Jackson, J.M., (1993). Critical review of behavioral interventions to preserve the environment. Research since 1980. *Environment and*

Behavior, 25, 275-321.

- Dougher, M. J. (1994). The act of acceptance. In S. C. Hayes, N.S. Jacobson, V. M. Follette, & M. J. Dougher (Eds.), *Acceptance and change: Content and context in psychotherapy*. Reno, NV: Context Press.
- Everson, H. T., and Tobias, S. (1998). The ability to estimate knowledge and performance in college: A metacognitive analysis. *Instructional Science*, 26, 65-79.
- Eze-Uzomaka, P., & Oloidi, J. A. (2017). Modernization and its effect on cultural heritage in southwestern Nigeria. AFRREV IJAH: *An International Journal of Arts and Humanities*, 6(2), 81-93.
- Fischer, P. A. & Malmberg, G. (2001). Settled people don't move: on life course and (im-)mobility in Sweden. *International Journal of Population Geography*, *7*, 357-371.
- Fischhoff, B., and Kadvany, J. (2011). *Risk: A Very Short Introduction*. Oxford. UK: Oxford University Press.
- Frewer LJ, Scholderer J, Bredahl L. (2003). Communicating about the risks and benefits of genetically modified foods: The mediating role of trust. *Risk Analysis 23*(6): 1117–1133 [PubMed] [Google Scholar].
- 藤井聡 (2003). 社会的ジレンマの処方箋-都市・交通・環境問題のための心理学, ナカニシャ 出版.
- Fujii, S. (2008). Doboku-keikakugaku. Kyoto: Gakugei Shuppansha. [Japanese].
- Fujii, S. (2006). Environmental concern, attitude toward frugality, and ease of behavior as determinants of pro-environmental behavior intentions. *Journal of Environmental Psychology*, 26(4), 262–268. doi:10.1016/j.jenvp.2006.09.003.
- Gärling T, Jakobsson C, Loukopoulos P, Fujii S (2008). Acceptability of road pricing. Book.
- Greenberg, J. (1990). Looking fair versus being fair: Managing impressions of organizational justice. In B. Staw & L. Cummings (Eds.), *Research in organizational behavior* (Vol. 12, pp. 111-157), Greenwich, CT: JAI.
- Godschalk, D.R., & Paterson, R.G., (1999). Collaborative conflict management comes of age. *Journal* of Architectural and Planning Research, 16(2): 91-95.
- Guggenheim, S., Wiranto, T., Prasta, Y. & Wong, S., (2004). *Indonesia Kecamatan Development Program: A Large-Scale Use of Community Development to Reduce Poverty*. Jakarta: The World Bank.
- Gyda Maras Sindre, (2012). Civic Engagement and Democracy in Post-Suharto Indonesia: A Review of Musrenbang, the Kecamatan Development Programme, and Labour Organising. *PCD Journal Vol. IV* No. 1 2012.
- Hatori, T., & Kajiwara, K. (2012). Protected values on public works and their acceptance. *Journal of Japan Society of Civil Engineers (Infrastructure Planning and Management)*, 68, I_231-I_239.
- Hatori, T., & Kajiwara, K. (2014). An effect of providing opportunities for reflection on the changes in

protected values: Implications for consensus building around public works. *Journal of Human Environmental Studies*, 12, 105-111.

- 羽鳥剛史,梶原一慶 (2014). 内省機会の提供が保護価値の変容に及ぼす影響―公共事業合意形成問題への示唆―,人間環境学研究, 12(2), 105-111.
- 川村祥代,川本義海 (2008). HLW最終処分場の誘致事例における課題, 土木計画学研究・講演 集, 38, CD-ROM.
- Hayes, S. C. (1994). Content, context, and the types of psychological acceptance. Chapter in Hayes, S. C., Jacobson, N. S., Follette, V. M. & Dougher, M. J. (Eds.), *Acceptance and change: Content and context in psychotherapy* (pp. 13-32). Reno, NV: Context Press.
- Hamersma, M, Heinen, E, Tillema, T et al. (1 more author) (2016). Residents' responses to proposed highway projects: Exploring the role of governmental information provision. *Transport Policy*, 49. pp. 56-57. ISSN 1879-310X.
- Heaton, T., Fredrickson, C., Fuguitt, G. V., & Zuiches, J. J. (1979). Residential preferences, community satisfaction, and the intention to move. *Demography*, *16* (4), 565-573.
- Hernández, B., Carmen, H. M., Salazar-Laplace, M. E., & Hess, S. (2007). Place attachment and place identity in natives and non-natives. *Journal of Environmental Psychology*, 27(4), 310–319.
- Insch, A., & Florek, M. (2008). A great place to live, work and play: Conceptualising place satisfaction in the case of a city's residents. *Journal of Place Management and Development*, 1(2), 138–149.
- Kearns, A. & Mason, P. (2013). Defining and measuring displacement: is relocation from restructured neighbourhoods always unwelcome and disruptive? *Housing Studies*, 28(2), 177-204.
- Kim, J. H., Pagliara, F., & Preston, J. (2005). The intention to move and residential location choice behavior. Urban Studies, 42 (9), 1621-1636.
- Kim, J., Schmöcker, J. D., Fujii, S., & Noland, R. B. (2013). Attitudes towards road pricing and environmental taxation among US and UK students. *Transportation Research Part A: Policy and Practice*, 48, 50–62.
- Kühberger, A., Schulte-Mecklenbeck, M., & Perner, J. (2002). Framing decisions: Hypothetical and real. *Organizational Behavior and Human Decision Processes*, *89*, 1162-1175.
- 桑子敏雄 (2011). 社会基盤整備での社会的合意形成のプロジェクト・マネジメント, 猪原健弘 (編) 合意形成学, 勁草書房, 179-202.
- 大沼進, 佐藤浩輔, 北梶陽子, 石山貴一 (2015). NIMBYを巡る当事者性の違いによる認識の差 と手続き的公正の保護価値緩和効果: 幌延深地層研究センターを題材としたシナリオ調査, 日本リスク研究学会誌, 25(3), 121-130.
- 大友章司,大澤英昭,広瀬幸雄,大沼進 (2014). 福島原子力発電所事故による高レベル放射性廃 棄物の地層処分の社会的受容の変化,日本リスク研究学会誌, 24(1), 49-59.
- Lind, E. A., & Tyler, T. R. (1988). The Social Psychology of Procedural Justice. New York: Plenum.

- Lim, C. S. & Baron, J. (1997). *Protected values in Malaysia, Singapore, and the United States*. (Unpublished manuscript), University of Pennsylvania.
- Lynn J. Frewer C., Howard R. S. (1998). The influence of initial attitudes on responses to communication about genetic engineering in food production. *Agriculture and Human Values Volume 15*, Issue 1, pp 15–30.
- Matha Bicket, Robin Vanner, (2016). Designing Policy Mixes for Resource Efficiency: The Role of Public Acceptability. *Sustainability*, *8*, 366; doi:10.3390/su8040366.
- Marije Hamersma, Eva Heinen, Taede Tillema and Jos Arts. (2016). Residents' responses to proposed highway projects: Exploring the role of governmental information provision. *Transport Policy*, vol. 49, issue C, 56-67.
- McDonald-Wilmsen, B., & Webber, M. (2010). Dams and displacement: raising the standards and broadening the research agenda. *Water Alternatives*, *3*(2), 142–161.
- Mehta, P. (2005). Internally-displaced persons and the Sardar Sarovar project: a case for rehabilitative reform in rural media. *American University International Law Review*, *20*(3), 613–647.
- Menhas, R., Umer, S., Akhtar, S., & Shabbir, G. (2015). Impact Of Modernization On Religious Institution: A Case Study Of Khyber Pakhtun Khwa, Pakistan. *European Review Of Applied Sociology*, 8(10), 23-28.
- Morgan, P. (2010). Towards a developmental theory of place attachment. *Journal of Environmental Psychology*, *30*(1), 11–22. https://doi.org/10.1016/j.jenvp.2009.07.001.
- Payne, J. W., Bettman, J. R., & Luce, M. F. (1996). When time is money: Decision Behavior under opportunity-cost time pressure. *Organizational Behavior and Human Decision Processes*, 66(2), 131–152. doi:10.1006/obhd.1996.0044.
- Pemberton, J., (1994). On the Subject of "Java". Ithaca, N.Y.: Cornell University Press.
- Putnam, R., Leonardi, R. & Nanetti, R., 1993. Making Democracy Work: Civic Traditions in Modern Italy. Princeton, N.J.: Princeton University Press.
- Rozenblit, L., and Keil, F. (2002) The misunderstood limits of folk science: An illusion of explanatory depth, *Cognitive Science*, *26*, 521-562.
- 坂本修一,神田啓治 (2002). 高レベル放射性廃棄物処分地選定の社会的受容性を高めるための 課題に関する考察,日本原子力学会和文論文誌,1(3),18-29.
- 曽我部蓮,羽鳥剛史 (2013). 公共事業における保護価値の規定要因に関する研究, 平成24年度 土木学会四国支部技術研究発表会原稿.

三宮真智子 (2008). メタ認知, 北大路書房.

- Sarosa, W. et al. (2005). Making People's Voices Matter: An Analytical Study on District Planning and Budgeting. Jakarta: Badan Perencanaan Pembangungan Nasional (Bappenas) & Decentralisation Support Facility (DSF).
- Schade, J., & Schlag, B. (2003). Acceptability of urban transport pricing strategies. Transportation

Research Part F: Traffic Psychology and Behaviour, 6(1), 45-61.

- Schmöcker, J.-D., Pettersson, P., & Fujii, S. (2012). Comparative analysis of proximal and distal determinants for the acceptance of coercive charging policies in the UK and Japan. *International Journal of Sustainable Transportation*, 6(3), 156–173.
- Schwartz H. S., Bardi A. (2001). Value hierarchies across culture, *Journal of Cross-Cultural Psychology 32*(3):268-290 · May 2001 with 1,141 Reads DOI: 10.1177/0022022101032003002.
- Sidel, J.T., (2007). *Riots, Pogroms, Jihad: Religious Violence in Indonesia. Singapore*: Singapore University Press.
- Skitka, L. J. (2002). Do the means always justify the ends, or do the ends sometimes justify the means?
 A value protection model of justice reasoning. *Personality and Social Psychology Bulletin, 28*, 452–461.
- Skitka, L. J., & Mullen, E. (2002). Understanding judgments of fairness in a real-world political Context: A test of the value protection model of justice reasoning. Personality and Social Psychology Bulletin, 28(10), 1419–1429. doi:10.1177/014616702236873.
- Stephens, M. (2010). Locating Chinese urban housing policy in an international context. *Urban Studies*, 7(14), 2965–2982.
- Sunstain, C. R. (2000). Deliberative trouble? Why groups go to extremes. *Yale Law Journal*, *110*, 71-119.
- Tetlock, P. E., Kristel, O. V., Elson, S. B., Green, M. C., and Lerner, J. S. (2000). The psychology of the unthinkable: Taboo trade-offs, forbidden base rates, and heretical counterfactuals. *Journal of Personality and Social Psychology*, 78, 853-870.
- Thibaut, J. W., & Walker, L. (1975). *Procedural justice: A psychological analysis*. L. Erlbaum Associates.
- Tyler, T., Degoey, P., & Smith, H. (1996). Understanding why the justice of group procedures matters: A test of the psychological dynamics of the group-value model. *Journal of Personality and Social Psychology*, *70*(5), 913.
- Tyler, T. R., Rasinski, K. A., & Spodick, N. (1985). Influence of voice on satisfaction with leaders: Exploring the meaning of process control. *Journal of Personality and Social Psychology*, *48*(1), 72.
- Van Voorst, R. (2015). Applying the risk society thesis within the context of flood risk and poverty in Jakarta, Indonesia. *Health, Risk & Society*, 17(3–4), 246–262.
- Vollmer, D., Prescott, M. F., Padawangi, R., Girot, C., & Grêt-Regamey, A. (2015). Understanding the value of urban riparian corridors: considerations in planning for cultural services along an Indonesian river. *Landscape and Urban Planning*, 138, 144–154.
- Yeung, Y. M. & Wong, T. K. Y. (2003). *Fifty Years of Public Housing in Hong Kong* (Hong Kong: Hong Kong Housing Authority).
- Wan, C., Shen, G. Q., & Choi, S. (2017). A review on political factors influencing public support for urban environmental policy. *Environmental Science and Policy*, 75, 70-80.

- Wenzel Michael., (2002). The Impact of Outcome Orientation and Justice Concerns on Tax Compliance: The Role of Taxpayers' Identity. *Journal of Applied Psychology* 87(4):629-45 · September 2002 with 333 ReadsDOI: 10.1037/0021-9010.87.4.629.
- Zverinova Iva, Milan Scasny, Eva Kysela, (2014). *What Influences Public Acceptance of the Current Policies to Reduce GHG Emissions? Technical Report* number: WP2 Affiliation: Prague, Charles University Environment Center.

Table of Contents

CHAPTER 1	1
Introduction	2
1.1. Modernization and trends in infrastructure development in Indonesia	2
1.2. Democratization and citizen participation in Indonesia	3
1.3. Public acceptance of infrastructure projects	5
1.4. Social conflict around infrastructure projects	6
1.5. Protected values	7
1.6. Purpose of this research	8
1.7. Structure of the Dissertation	8
CHAPTER 2	11
Theoretical Background	11
2.1. Determinants of acceptance	11
2.2. Properties of protected value	12
2.3. Procedural fairness	14
2.4. The focus of present research	16
CHAPTER 3	18
The Effects of Protected Values on Policy Acceptance	18
3.1. Introduction3.1.1. Trade-off judgment of relocation and protected values	18 18 20
3.2. Method 3.2.1. The study area and respondents	21 21
3.3.Questionnaire	23
3.4. Results	25
CHAPTER 4	28
The Impact of Protected Values on Acceptance and Procedural Fairness of Public Projects Exploring the Effects of Decision Procedures	3: 28
4.1. Introduction	28
4.1.1. Protected Values	28
4.1.2. Protected values and decision procedures	30 31
4.2. Method	31
4.2.1. Participants	31 גר
4.3 Results	34
4.3.1. Prevalence and Properties of Protected Values	

4.3.2. Acceptance associated with protected values and decision rules4.3.3. Assessments of decision procedures	35 36
4.4. Discussion	
CHAPTER 5	39
Protected Values and Their Influences on Perceived Procedural Fairness in Highw A Field Survey in South Sulawesi, Indonesia	ay Project: 39
5.1. Introduction	
5.2. Theory	40
5.2.1. Properties of protected values	40
5.2.2. Protected values and procedural fairness	41
5.3. Methods	43 43 45
5.4. Procedures and instruments	45
5.5. Results	47
 5.5.1. Possession and properties of protected values	47 48 49 50 51
5.6. Discussion	52
CHAPTER 6 Protected Values and Illusion of Understanding Around Risk Acceptance: A Vignette Experiment of the Construct of High Level Radioactive Waste	55 55 55
 6.1. Introduction 6.1.1. The concept of protected values 6.1.2. Protected values and the illusion of understanding 6.1.3. Hypothesis of the present study	
6.2. Experiment.6.2.1. Experimental participants.6.2.2. Experimental procedures.	59 59 60
 6.3. Results	
6.4. Discussion6.4.1. Implications for the issue of consensus formation6.4.2. Subjects for future investigation	65 66 68
CHAPTER 7	70
General Conclusions	70

List of Tables

Table 3. 1. Questions used to measure each variable	24
Table 3. 2. Estimates of coefficients in measurement models	26
Table 4. 1. Personal attributes of participants 32	
Table 4. 2. Scenario for decision procedures	34
Table 4. 3. Characteristics of deontological rules associated	35
Table 4. 4. Assessments of decision procedures for participants with protected values	37
Table 5.1. Odds ratios for possession of protected values (n=300).	48
Table 5. 2. Means of post-perceptions of procedural fairness associated	50
Table 5. 3. Total effects (standardized of coeficients)	52
Table 6. 1. Measurements of the psychological characteristics of protected values	61
Table 6. 2. Distribution of the possession of protected values in two surveys	63
Table 6. 3. Evaluations of self-assessments of understanding in two surveys	63

List of Figures

Figure 3. 1.A theoretical model of determinants of acceptance of relocation program	21
Figure 3. 2 Map of Jakarta province	22
Figure 3. 3 Research area	22
Figure 3. 4.(a) Housing conditions, (b) riverbank settlements, and (c) post-flood inundation	l
damage	23
Figure 3. 5. Estimated model of determinants of acceptance of relocation program. (The	
estimated	25
Figures 4. 1. Acceptance associated with protected values	36
Figure 5. 1. Survey area	44
Figure 5. 2. The image of the Simpang 5 project	44
Figure 5. 3. Means of the properties of the deontological rule among people	49
Figure 5. 4. Means of perceptions of pre- and post-procedural fairness among	50
Figure 6. 1.Flow chart of the experiment	60
Figure 6. 2. Means of self-assessments of understanding before and after the experimental.	64
Figure 6. 3. Means of self-assessments of confidence before and after the experimental	65